

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2020JG006165

Special Section:

Winter limnology in a changing world

E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, and S. Sadro are co-first authors.

Key Points:

- Many lakes experience ice cover, yet there are no general frameworks for understanding responses to differing ice and snow conditions
- The Lake Ice Continuum Concept is a framework representing how lake ecosystems vary along a continuum of energy inputs mediated by winter
- We present a new approach for characterizing lakes within a landscape context and for testing our knowledge of changing winter climates

Correspondence to:

G. P. Rue, garrett.rue@colorado.edu

Citation:

Cavaliere, E., Fournier, I. B., Hazuková, V., Rue, G. P., Sadro, S., Berger, S. A., et al. (2021). The Lake Ice Continuum Concept: Influence of winter conditions on energy and ecosystem dynamics. *Journal of Geophysical Research: Biogeosciences*, *126*, e2020JG006165. https://doi.org/10.1029/2020JG006165

Received 3 FEB 2021 Accepted 15 OCT 2021

Author Contributions:

Conceptualization: E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, S. Sadro, S. A. Berger, J. B. Cotner, H. A. Dugan, S. E. Hampton, N. R. Lottig, B. C. McMeans, T. Ozersky, S. M. Powers, M. Rautio, C. M. O'Reilly Data curation: S. A. Berger, H. A. Dugan, M. Rautio Funding acquisition: S. E. Hampton, S. M. Powers Investigation: E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, S. Sadro, S. A. Berger, J. B. Cotner, H. A. Dugan, B. C. McMeans, T. Ozersky, S. M. Powers, M. Rautio Project Administration: C. M. O'Reilly

© 2021. American Geophysical Union. All Rights Reserved.

The Lake Ice Continuum Concept: Influence of Winter Conditions on Energy and Ecosystem Dynamics

E. Cavaliere¹ ^(D), I. B. Fournier² ^(D), V. Hazuková³ ^(D), G. P. Rue⁴ ^(D), S. Sadro⁵ ^(D), S. A. Berger⁶ ^(D), J. B. Cotner⁷ ^(D), H. A. Dugan⁸ ^(D), S. E. Hampton⁹ ^(D), N. R. Lottig¹⁰ ^(D), B. C. McMeans¹¹ ^(D), T. Ozersky¹² ^(D), S. M. Powers¹³ ^(D), M. Rautio^{14,15,16} ^(D), and C. M. O'Reilly¹⁷ ^(D)

¹Global Institute for Water Security, University of Saskatchewan, Saskatoon, SK, Canada, ²Département de Biologie, Université Laval, Québec City, QC, Canada, ³School of Biology and Ecology, Climate Change Institute, University of Maine, Orono, ME, USA, ⁴Institute for Arctic and Alpine Research, University of Colorado Boulder, Boulder, CO, USA, ⁵Department of Environmental Science and Policy, University of California, Davis, Davis, CA, USA, ⁶Department of Experimental Limnology, Leibniz-Institute of Freshwater Ecology and Inland Fisheries (IGB), Stechlin, Germany, ⁷Department of Ecology, Evolution and Behavior, University of Minnesota, Twin Cities, St Paul, MN, USA, ⁸Center for Limnology, University of Wisconsin–Madison, Madison, WI, USA, ⁹Center for Environmental Research, Education and Outreach, Washington State University, Pullman, WA, USA, ¹⁰Center for Limnology Trout Lake Station, University of Wisconsin–Madison, Boulder Junction, WI, USA, ¹¹Department of Biology, University of Toronto Mississauga, Mississauga, ON, Canada, ¹²Large Lakes Observatory, University of Minnesota Duluth, Duluth, MN, USA, ¹³Department of Biology, Baylor University, Waco, TX, USA, ¹⁴Département de Sciences Fondamentales, Université du Québec à Chicoutimi, Saguenay, QC, Canada, ¹⁵Center for Northern Studies (CEN), Université Laval, Québec City, QC, Canada, ¹⁶Group for Interuniversity Research in Limnology and Aquatic Environment (GRIL), Université de Montréal, Montreal, QC, Canada, ¹⁷Department of Geography, Geology, and the Environment, Illinois State University, Normal, IL, USA

Abstract Millions of lakes worldwide are distributed at latitudes or elevations resulting in the formation of lake ice during winter. Lake ice affects the transfer of energy, heat, light, and material between lakes and their surroundings creating an environment dramatically different from open-water conditions. While this fundamental restructuring leads to distinct gradients in ions, dissolved gases, and nutrients throughout the water column, surprisingly little is known about the resulting effects on ecosystem processes and food webs, highlighting the lack of a general limnological framework that characterizes the structure and function of lakes under a gradient of ice cover. Drawing from the literature and three novel case studies, we present the Lake Ice Continuum Concept (LICC) as a model for understanding how key aspects of the physical, chemical, and ecological structure and function of lakes vary along a continuum of winter climate conditions mediated by ice and snow cover. We examine key differences in energy, redox, and ecological community structure and describe how they vary in response to shifts in physical mixing dynamics and light availability for lakes with ice and snow cover, lakes with clear ice alone, and lakes lacking winter ice altogether. Global change is driving ice covered lakes toward not only warmer annual average temperatures but also reduced, intermittent or no ice cover. The LICC highlights the wide range of responses of lakes to ongoing climate-driven changes in ice cover and serves as a reminder of the need to understand the role of winter in the annual aquatic cycle.

Plain Language Summary Millions of lakes worldwide freeze during winter. The formation of lake ice dramatically alters lakes by isolating them from their surrounding landscape and atmosphere. The thickness and optical qualities of ice and snow regulate the amount of solar radiation entering lakes, while shielding them from wind energy. Consequently, lake ice is an important factor regulating physical mixing dynamics within lakes, and structuring vertical thermal and chemical gradients. Although organisms from bacteria to fish have adapted to the winter environment, we lack a comprehensive understanding of how variation in lake ice cover conditions affects fundamental ecosystem processes or food web structure. Here, we combine a synthesis of the current literature with three novel case studies to develop the Lake Ice Continuum Concept as a framework for understanding how key aspects of the physical, chemical, and ecological structure and function of lakes vary along a continuum of energy inputs mediated by winter climate. This framework is useful for understanding changes associated with seasonal ice dynamics and for predicting how lakes may respond to climate change.



Supervision: E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, S. Sadro, C. M. O'Reilly

Visualization: E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, S. A. Berger, H. A. Dugan, S. M. Powers, C. M. O'Reilly

Writing – original draft: E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, S. Sadro, S. A. Berger, J. B. Cotner, H. A. Dugan, N. R. Lottig, B. C. McMeans, T. Ozersky, S. M. Powers, M. Rautio, C. M. O'Reilly

Writing - review & editing: E.

Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, S. Sadro, S. A. Berger, J. B. Cotner, H. A. Dugan, S. E. Hampton, N. R. Lottig, B. C. McMeans, T. Ozersky, S. M. Powers, M. Rautio, C. M. O'Reilly

1. Introduction

Lake ice has profound ecological and societal consequences (Denfeld et al., 2018; Knoll et al., 2019; Özkundakci et al., 2016; Sharma et al., 2019). There are an estimated 117 million lakes worldwide, most of which are located above 45° North (Verpoorter et al., 2014) and typically freeze over during winter. Although our understanding remains limited in comparison to studies conducted during ice-free seasons, numerous studies have documented the physical, chemical, and biological processes that occur under ice (e.g., Bižić-Ionescu et al., 2014; Denfeld et al., 2018; Hampton et al., 2017; McMeans et al., 2020; Powers, Labou, et al., 2017; Rue et al., 2019). Despite often being considered a period of relative quiescence, ecosystem functions during the winter are often dynamic (Bertilsson et al., 2013; Denfeld et al., 2018; Grosbois et al., 2017; Jansen et al., 2021). Moreover, there is growing evidence that winter conditions set the stage for those of summer, and vice versa (Hampton et al., 2017), and that lakes must be viewed across a full seasonal continuum (Salonen et al., 2009).

A comprehensive understanding of winter ecosystem structure and function must include the full range of interannual and long-term variation experienced by a lake. While some lakes experience little or no snow cover (Schindler & Smol, 2006) or interannual variability in snow cover, warmer winters are likely to cause increasing ablation of the snowpack and precipitation falling as rain rather than snow in the future (Dumanski et al., 2015; Fernandez et al., 2020; Sadro et al., 2018; Sickman et al., 2003). Many lakes with extended records of ice history show declines in ice duration over the last century (Benson et al., 2012; Magnuson, 2000), a trend that is expected to continue with climate warming (Caldwell, Chandra, Albright, et al., 2020; Filazzola et al., 2020). The number of lakes periodically experiencing an ice-free winter is expected to increase by hundreds of thousands depending upon the intensity of warming (Sharma et al., 2019). Climate warming is also impacting lakes in polar regions, where thinning margins of perennially ice covered lakes are creating new hotspots for littoral productivity (Bégin et al., 2021; Castendyk et al., 2016). Although changes in lake ice conditions appear to be accelerating (Imrit & Sharma, 2021) and may have a myriad of cascading effects (Preston et al., 2016), we lack a conceptual framework to understand and predict the effect of such changes on ecosystem structure and function.

Many of the effects of ice cover arise from the constraints it places on energy fluxes associated with transmission of solar radiation, mixing dynamics, and heat exchange. Depending on the lake and year, the duration of seasonal ice cover can range from as short as a few days to the entire year, and ice thickness from less than a centimeter to multiple meters. The formation of lake ice severs or mutes linkages with the landscape and atmosphere, creating an aquatic environment that strongly contrasts with open-water conditions. Viewed in terms of light inputs and connectivity to the atmosphere/landscape (Figure 1a), the response of ice-covered lakes to seasonal changes in light and energy inputs resemble hysteresis loops. Light and energetic inputs in the winter are low, increasing during spring and summer until lakes become ice-free, with the shape and lag of these trajectories modulated by ice cover dynamics. Changes in climate that reduce the duration of ice cover, cause intermittency, or result in its complete loss have the potential to dramatically alter winter light and thermal regimes within lakes; effects that can also carry over to subsequent seasons (Figure 1b). Consequently, developing a general framework is imperative for characterizing how the ecological structure and function of lakes may shift under future climate scenarios. Such a framework is equally important for understanding differences among lakes that span large gradients in winter ice dynamics. Drawing from a synthesis of the literature, we present a conceptual model (Figure 2) that explores these conditions as a continuum of lake ice duration and condition (e.g., thickness and snow depth), which regulate energy inputs and subsequent physical, chemical, and biological structure. In support, we offer winter case study examples from three lakes (Simoncouche, Mendota, and Stechlin; Boxes 1-3) to demonstrate key processes and patterns relevant to the conceptual model, exceptions thereto, and associated hypotheses in a changing world.

2. The Lake Ice Continuum Concept

The Lake Ice Continuum Concept (LICC) provides a framework for understanding how lake ecosystems vary along a continuum of energy inputs mediated by winter climate, ice, and snow. The LICC is not intended to represent a continuum of successive states through which an individual lake will necessarily transition in association with changes in climate. In lakes that experience seasonally freezing conditions, the ice and snow regime a given





Figure 1. Hysteresis patterns of average annual light and connectivity conditions in a given lake under (a) Historical & Current Lake Seasonality and (b) Future Lake Trajectories under Climate Warming as lakes experience a loss of snow and ice cover with reductions in the duration of winter. The shifts in seasonality further alters the lake temperature regime, reducing the period of colder temperatures (blue) and prolonging warmer conditions (yellow-orange-red). Lake thermal regimes and the responses of individual lakes to climate warming will in part depend on factors such as lake depth, morphometry, and water clarity.

lake experiences will be regulated by a combination of large and small scale factors. Climate variability along latitudinal and elevational gradients interacts with local scale factors such as aspect, land cover, lake morphometry, and lake trophic state to drive the energetics affecting seasonal and interannual variability in ice and snow. These properties may also mediate the lake ecosystem response to changing winter conditions. Variation in ice thickness, ice structure or optical quality, depth of snow cover, and overall duration of ice cover together regulate exposure to wind, incoming solar radiation, and other atmospheric/landscape processes, and establish the dominant mixing regimes within lakes (Figure 2.1). The presence of ice and the climatic conditions that cause precipitation to fall as snow reduce the connectivity between lakes and their air- and watersheds (Figures 2.1a and 2.1b). Periods without winter ice cover have the potential for higher energetic fluxes through wind exposure, and higher or altered input rates of allochthonous material when precipitation falls as rain (Figure 2.1c). As a result of the increased connectivity to the atmosphere and landscape, lakes that lose ice cover will respond more directly to variation in winter or early spring weather patterns (Smits et al., 2020), leading to profound differences in the early spring lake condition.

By regulating energetic fluxes, lake ice determines the dominant physical mixing regimes within lakes (Lewis, 1983), influences time scales of mixing, controls the distribution of matter and energy throughout the water column, and establishes the template for biological structure and ecosystem function (Figure 2.2). Thus, the duration of ice cover also factors in the development and strength of redox and biogeochemical gradients in the water column. Differences in phytoplankton biomass and rates of primary productivity along the continuum of light availability in ice-covered lakes will drive differences in the oxycline and variation in redox potential with depth. Such vertical gradients will in turn affect biochemical reactions and, together with differences in rates of biological uptake, the distribution of nutrients (Figures 2.2 and 2.3b). Variation in some aspects of ecosystem structure along the LICC may be approximately linear, such as with water column stability or redox potential. Others may involve step changes, as with the change in mixing depth (z_m) and euphotic depth (z_{su}) among lakes with or without ice (Figure 2.3b), or changes in secondary production fueled by increased phytoplankton biomass in lakes with clear ice compared to lakes with ice and snow (Figure 2.3c). Although heterotrophic microbial processes will continue across the entire lake ice continuum, there will be fundamental differences in microbial metabolism and community composition under aerobic versus anaerobic conditions, as a function of complex interactions among water temperature, organic substrate, nutrient, and light availability (Dillon et al., 2020; Tran et al., 2018; Vigneron et al., 2019). We predict that the importance of microbial production for sustaining food webs will be highest under conditions where ice and snow persist and decline with increasing energy inputs and mixing along the LICC (Figure 2.3c),





Figure 2. Lake Ice Continuum Concept: framework for understanding how conditions vary along a continuum of energy inputs mediated by winter climate, conditions mediated by ice, and snow in a representative mesotrophic temperate lake of moderate size and depth, with at least one annual full mixing. (1) We highlight how lake ecosystem dynamics change when winter ice cover is (a) snow-covered (dotted gray), (b) clear (light blue), or (c) absent (dark blue). Moving across a continuum of winter energetic constraints imposed by variation in ice and snow, lakes will experience increased inputs of energy and allochthonous material from the atmosphere and the surrounding landscape. Increasing energy inputs into the lakes will destabilize the water column (arrows and swirls) and affect the distribution of gases, particles, and heat. Also, light penetration will deepen as lake ice becomes clearer or completely disappears (yellow arrows). (2) Across the continuum, we show vertical shifts in the following variables: (a) mixing depth (z_m), (b) temperature (Temp), (c) depth of the photic zone (z_{eu}), (d) photosynthetically active radiation (PAR), (e) dissolved oxygen saturation (DO), (f) chlorophyll-*a* (chl-*a*), (g) pH, (h) greenhouse gases (GHGs), (j) specific conductance (SC), (k) total phosphorus (TP), (l) nitrate (NO₃), and (m) ammonium (NH₄). (3) We show (a) primary, (b) physical and chemical, and (c) biological responses along the continuum of winter conditions. Increasing width of wedges suggests an increase in response variable and wedge color corresponds to the shift along the continuum of winter conditions.

but recognize other factors, such as increased lability of dissolved organic matter (DOM) from microbial or phytoplankton production, could mediate this general pattern. Likewise, we predict an increase in heterotrophic, chemolithotrophic, phagotrophic, or mixotrophic strategies among algae and microbes along a gradient of decreasing light (Özkundakci et al., 2016; Rautio et al., 2011). Among higher trophic levels, we expect increased light under clear ice or no ice conditions to benefit zooplankton, benthic organisms, and fish due to increased primary production and visual prey detection (Varpe et al., 2015). Higher rates of secondary production and foraging success are expected to increase the number and strength of trophic linkages within food webs (Figures 2.3a and 2.3c; Perga et al., 2020).



Box 1: Responses to intra-annual variation in snow cover in Lake Simoncouche

Lake Simoncouche (48°13'52"N 71°15'3"W) is a dimictic, mesotrophic lake in Québec (Canada) with a surface area of 0.83 km², 8 m maximum depth, and 2.5 m mean depth. The lake is usually ice-covered from mid-November to early May, with ice thickness reaching up to 1.2 m in some years. Measurements of ice and snow coverage were collected manually during the winter of 2017-2018, with a nearly complete loss of snow in January caused by a rain-on-snow event. A sensor system installed at 1.4 m continuously measured temperature, specific conductance, and dissolved oxygen saturation. Data available from Rautio et al., 2021.

The variability in snow and ice thickness during winter caused shifts in photosynthetically active radiation (PAR), temperature, specific conductance (SC), and dissolved oxygen saturation (DO). The onset of ice cover in late-November attenuated PAR; however, greater reductions were linked to the presence of snow cover on ice. A snow thickness of less than 15 cm was enough to virtually block PAR transmission. In spring, PAR was not detectable until after the loss of snow, and then increased substantially after ice-off and as solar radiation increased in May. After ice-on, water temperature at 1.4 m was stable and appeared insensitive to variability in ice and snow conditions. In May, when ice began to melt, temperature initially dropped as cold meltwater mixed with lake water, but later increased responding to higher radiant heat flux. SC rose steadily throughout the winter. likely as a result of cryo-concentration under the increasing ice thickness and potentially diffusive processes, suggesting minimal mixing. In April, as the ice thickness decreased, meltwater diluted the ions, decreasing SC. When ice coverage began to disappear and the lake began to mix, SC increased again Draw-down of DO that lasted until late April was possibly the result of respiration and decomposition of organic matter. In Lake Simoncouche, macrophytes cover more than 25% of the lake surface area; hence, they can be a major source of particulate and dissolved organic matter, fueling respiratory processes under ice. In late April, increasing PAR enhanced primary productivity, and together with lateral flows of oxygenated water, increased DO



Fig. 1: Lake Simoncouche time-series data of (a) mean in-situ daily photosynthetically active radiation (PAR) values, biweekly snow (grey), and ice thickness (light blue) measurements, (b) high-frequency temperature (grey), specific conductance (brown), and dissolved oxygen saturation (blue) collected during the winter of 2017-2018.

3. Ecosystem Dynamics Along the Lake Ice Continuum

To illustrate the LICC, we examine three possible winter conditions along the lake ice continuum, which differ in energetic constraints: (a) lake ice with snow cover (constraints to solar radiation penetration, wind-mixing and atmospheric/landscape exchanges), (b) clear ice with no snow (more solar radiation inputs, but constraints to wind-mixing and atmospheric/landscape exchanges), and (c) no ice (all constraints lifted, Figure 2). For each of these scenarios, we characterize the dominant physical processes and synthesize what is known about how differences in energetics affect chemical and biological dynamics. While our discussion is categorical, individual lake winter conditions along the continuum may vary intra- and inter-annually (Beall et al., 2015; Prowse & Stephenson, 1986). For example, ice can form and melt multiple times, snowfall can accumulate or be swept clear of lake ice by winds, ice quality can change throughout the winter, and the duration of snow-covered ice versus clear ice may vary among lakes and years-all affecting surface albedo and ecosystem energetics. Consequently, conditions of individual lakes will depend on where they fall along the continuum in ice cover and snow cover through time within any given winter, or through multiple winters across years (Beall et al., 2015; Bruesewitz et al., 2015).

The LICC may be applied to any lake that experiences annual to seasonal periods of ice-cover. Although we describe the LICC as it applies to a representative lake (i.e., mesotrophic lake of moderate size and depth that is ice-free for some period of the year, mixing at least once; Figure 2), individual lake responses to changing winter conditions may be further affected by lake-specific characteristics such as trophic state or morphometry. For example, lake trophic state or lake size may interact with the lake ice continuum to mediate energy fluxes into lakes, affecting sediment heat fluxes and mixing, differentially fueling pelagic or benthic production, and subsequently influencing biotic responses and the fate of winter production within lake food webs. For this reason, in addition to our discussion of these three scenarios along the lake ice continuum, we present case studies from lakes that vary in size and trophic state, and explore how they deviate from expectations based on the LICC framework. Such empirical tests will be an important aspect of evaluating the applicability of the concepts underlying the LICC and predicting how individual lakes may respond to climate change. Ultimately, we hope that this concept







Fig. 1: Satellite imagery of ice covered Lake Mendota, Wisconsin shows stark differences between the condition of ice cover on March 15th, 2018 when the lake ice was almost entirely snow-free and on March 5th, 2019 when a layer of snow covered the ice. Both images have been processed to true color composites using RGB bands. While the dates of imagery do not overlap with the dates of manual sampling, the conditions represented in the satellite images reflect the conditions during which the vertical profiles presented in Fig. 2 were taken. Sentinel-2 multispectral imagery obtained courtesy of the European Space Agency.

Lake Mendota (43°06'24''N 89°25'29''W) is a dimictic, eutrophic lake in Madison, Wisconsin (USA) with a surface area of 39.6 km², 25 m maximum depth, and 12.5 m mean depth. Lake ice coverage usually lasts from mid-December to early-April. Under-ice light intensity was collected from a moored buoy approximately 1 m below the lake surface during Feb 26-28 in 2018 and throughout the month of February in 2019. Vertical water column profiles were taken on February 26th, 2018 and February 22nd, 2019 with a YSI EXO multiprobe. Although the total duration of ice cover was similar between the two years, total snow accumulation differed—in 2019 it was 137 cm, compared to 79 cm in 2018, as measured at a meteorological station northeast of Madison. Data available from Dugan, 2021b.

The effect of snow on radiative inputs can have a cascading influence on the lake ecosystem. The absence of snow dramatically increased the intensity of light transmitted, leading to water temperatures that were warmer by ~2°C throughout the entire water column. Antecedent conditions are important in setting up under-ice conditions, and while the water column of Lake Mendota is typically isothermal near 0.5°C prior to ice-on in December or January (Yang et al. 2021), in 2018 it was <1°C warmer soon after ice formation compared to 2019. Overall chlorophyll-a concentration was higher under clear ice, and the chlorophyll peak occurred deeper in the water column, likely as a result of greater light penetration and radiative conductive mixing This was associated with supersaturated oxygen conditions from the surface down to 10 m. The high DO saturation, along with higher pH under clear ice suggest that the increase in chlorophyll a led to higher primary production. In contrast, with snow on ice, chlorophyll-a maximum was right under the ice because of light-limitation and oxygen saturation was only about 75% in the upper 10 m. Specific conductance was generally lower with clear ice, reflecting potential differences in antecedent conditions



Fig. 2: Comparison of (a) light intensity at 1 m beneath the ice cover $(lm.m^2)$, from Feb 26-28 2018 (clear ice, blue) and Feb 1-28, 2019 (snow-covered ice, gray), and under-ice water column profiles in Lake Mendota from Feb 26, 2018 (blue) and Feb 22, 2019 (gray) of (b) water temperature, (c) chlorophyll-a, (d) dissolved oxygen saturation, (e) pH, (f) specific conductance.

tual model will be used to generate predictions and testable hypotheses to advance our understanding of winter limnology, and we encourage researchers to apply this framework across a spectrum of lakes.

3.1. Lake Ice With Snow Cover

When snow is present on ice, the lake environment has limited light availability and a stable water column. Snow on ice both increases albedo and limits transmission of solar radiation into the water column (Figure 2.1a). Mixing is limited to convective processes from sediment heat flux and minor turbulence at the surface due to internal waves from ice deformation by strain or wind stress. Inverse stratification persists (Figures 2.2b and 2.3b), with upward solute distribution by diffusion. Low light availability restricts the chlorophyll maximum to near the ice-water interface, further contributing to strong oxygen, redox, pH, and other biogeochemical gradients (Fig-



Box 3: Cross-seasonal linkages in years with cold, ice-covered winter, and warmer winter with no ice cover in Lake Stechlin

Lake Stechlin (53°09'20''N 13°01'52''E) is a deep, dimictic, oligo-mesotrophic, hardwater lake in the Baltic Lake District located in Northeast Germany. It has a surface area of 4.25 km², 69.5 m maximum depth, and 23.3 m mean depth. Lake ice coverage usually lasts from January to March, but in recent years, warm winters have led to shorter ice cover duration and even ice-free conditions throughout the winter. To demonstrate the effect of warmer, ice-free winters on cross-seasonal linkages, we selected two sets of profiles representing winter (February) and spring (April) conditions in two years with differing winters. Profiles were taken in 2012, the last year on record with a cold winter and full ice cover on the lake (February 15 when ice thickness = 15 cm and April 12, when the lake was ice-free), and in 2020, when winter conditions were warm and the lake was ice-free (February 5 and April 8, with the lake ice-free on both dates). A YSI 6000 multiprobe was used for vertical profiling in 2012 and YSI EXO in 2020 (intercalibrations of the instruments were performed). Water samples for nutrient analyses were collected at discrete depths and water clarity was measured with a Secchi disk (through a hole in the ice when the lake was ice-covered). Euphotic depth (z_{eu}) was then approximated using the following equation: $z_{eu} = 3x z_{secchi}$, assuming that a thin layer of clear ice does not substantially reduce light penetration. Data are available from Berger et al., 2021.

During the cold winter with clear ice cover in 2012, the water column was inversely stratified and temperature was approximately 2°C cooler relative to the warmer winter with no ice cover in 2020. The cooler water temperatures in 2012 persisted into spring, likely as a result of the buffering effect of lake ice cover on spring warming. While Lake Stechlin was dimicite in the year with ice cover (2012), it switched to monomixis when ice cover was absent (2020). As a result of warmer air temperatures during an ice-free winter, the lake did not lose enough heat throughout the fall and winter to mix completely, and full mixing was delayed until March 2020. The incomplete mixing during the ice-free winter subsequently had important implications for the distribution of solutes throughout the water column, resulting in the persistence of a large anoxic region below a depth of 40 m. Sustained anoxia in the hypolimnion fostered and prolonged the period of phosphorus loading from the sediments, leading to higher TP concentration in the spring following the warmer winter without ice cover as well as lower pH and higher hypolimnetic specific conductance. Weak gradients in the upper water column reflected deepening of the mixed layer and erosion of remaining stratification during winter with no ice cover. For example, chlorophyll-*a* was uniformly distributed throughout the mixed layer in the absence of ice cover; however, oxygen subsaturation (~70%) suggests active

mineralization of sinking, senescent organic matter. In contrast, during the cold winter when the lake was ice-covered, chlorophyll-a peaked at 1 m depth, corresponding to oxygen saturated conditions and indicating primary production just below the ice. While winter differences in oxygen saturation disappeared in the spring as the lake equilibrated with the atmosphere, chlorophyll-a remained higher (~3x) in spring following the cold winter with ice cover, possibly as a result of favorable light conditions in the upper water column after the onset of stratification ($z_{eu} = 20.1 \text{ m}$ in spring 2012). In contrast, water clarity was lower in spring following the warm winter without ice, shrinking the extent of the photic zone by more than 50% (zeu = 8.7 m in spring 2020). In both years, these differences in water clarity carried over from winter to spring (zeu = 15.9 m in 2012, and zeu = 10.5 m in 2020). During the year with warmer winter, the formation of an extensive surface-laver bloom of cyanobacterium Planktothrix rubescens (S. A. Berger, pers. obs., Berger et al., 2021) likely contributed to the decreased water clarity, creating light-limiting conditions for the phytoplankton distributed throughout the water column. Moreover, increased connectivity to the landscape and winter precipitation that fell as rain may have led to higher inputs of allochthonous material, further reducing light availability during the warm year



Fig. 1: Lake Stechlin winter (February, solid lines) and spring (April, dashed lines) profiles of (a)

temperature, (b) dissolved oxygen saturation, (c) total phosphorus, (d) pH, (e) specific conductance, and (f) chlorophyll-a from two years with different winter conditions. In 2012 the lake had a 15 cm thick, snow-free ice cover (light blue solid line), and in 2020 the lake was ice-free (dark blue solid line).

ure 2.2). Rates of both primary and secondary production are low, and microbial activity is expected to dominate, including feedback through the microbial loop (Figure 2.3c). However, there remain open questions regarding the role of the basal food web in subsidizing higher trophic levels during winter and across seasons.

Snow on ice reduces transmission of solar radiation, contributing to a stable inverse stratification (Figure 2.2b). The extent to which ice and snow reduce solar radiation depends in large part on the thickness of the snow and, to a lesser extent, the quality of the ice. Snow can reduce the transmissivity of light into lakes by over an order of magnitude (Box 1). Albedo can reach 80%–90% with just 5 cm of snow depth (Jakkila et al., 2009), and attenuation of remaining light by snow leads to higher transmissivity in the low-energy 500–600 nm range (Leppäranta, 2015; Perovich, 2007). In areas with heavy snowfall, the weight of snow can push lake ice into surface waters where it freezes as a slushy snow-ice matrix, further decreasing light penetration (Leppäranta, 2015; Melack et al., 2020). Because radiatively driven convective mixing near the surface is low, the majority of heat flux would come from heat stored in the sediments (Figure 2.1a). Such sediment heat fluxes can reach up to 5 W m⁻² (Lik-

ens & Ragotzkie, 1965; Malm, 1998), and can initiate density currents that flow downslope to the deepest part of the lake or reach neutral buoyancy (MacIntyre et al., 2018; Mortimer & Mackereth, 1958). Downslope flows can establish central upwelling and circulation cells with rates up to 10^{-5} mm s⁻¹ that persist until the sediment temperatures equilibrate with the water column (Kirillin et al., 2012). During the earlier phases of winter, sediment heat fluxes can be the main driver of mixing, depending upon lake depth and volume (Figure 2.1a; Jansen et al., 2021; MacIntyre et al., 2018).

Reduced light availability also depresses primary production and algal biomass, and respiration and decomposition are dominant (Figures 2.2f and 2.3c). In general, primary production is constrained to just below lake ice or shallow littoral zones (Marchand, 1987). However, depending upon light availability, chlorophyll-a maxima may not be immediately at the surface (e.g., Box 2). Considering that photosynthesis reaches saturation at lower irradiance at low temperatures, biochemical reactions may not be able to keep up with higher light intensity near the ice surface (Davison, 1991; Falkowski & Raven, 2007). While adaptations (e.g., increased concentration of enzymes, changing pigment composition and quantity) help phytoplankton cope with low-light and low-temperature conditions, we expect that snow-covered lakes will generally be net heterotrophic, as confirmed by oxygen depletion throughout the water column in many lakes (e.g., Brentrup et al., 2020; Catalan, 1992). The magnitude of heterotrophy is influenced by lake trophic status and catchment characteristics that regulate external carbon loads (Brentrup et al., 2020; Roiha et al., 2015). With primary production limited, the role of the microbial loop in transferring resources to higher trophic levels will be more prominent (Stockner & Porter, 1988). Mineralization may be an important process, particularly in smaller and more productive lakes where organic matter accumulates in the summer (Rabaey et al., 2021). However, there can be large differences in overwinter oxygen demand and extent of anoxia depending on water clarity and the relative contribution from pelagic and benthic production (Rabaev et al., 2021).

Together, stratification and limited oxygen production influence biogeochemical gradients. Snow covered lakes have strong gradients in dissolved oxygen (Figure 2.2e; Catalan, 1992), as well as vertical changes in pH (Figure 2.2g) and the concentrations and composition of DOM (Gonsior et al., 2013; Rue et al., 2019), greenhouse gases (GHGs; Figure 2.2h; Denfeld et al., 2018), nutrients, and other solutes (Figures 2.2j-2.2m; Cavaliere & Baulch, 2018, 2020; Joung et al., 2017; Rue et al., 2019). Oxygen consumption under ice is strongly influenced by transparency of the ice and snow (Granados et al., 2020), and these redox gradients facilitate microbial-driven redox reactions (Figure 2.3c), such as ammonification, nitrification (Cavaliere & Baulch, 2019; Powers, Baulch, et al., 2017), denitrification (Cavaliere & Baulch, 2018), sulfate reduction (Urban et al., 1994) and sulfide oxidation, organo-sulfogenesis (Rue et al., 2019), ferrotrophy, methanogenesis, and methanotrophy (Vigneron et al., 2019). Under-ice supersaturation of nitrogen gas and N_0O , both the products of denitrification and anammox (anaerobic ammonium $[NH_4]$ oxidation), suggest that concentrations of N (both NO₃ and NH₄) could be constrained by these microbial processes (Cavaliere & Baulch, 2018; Loeks-Johnson & Cotner, 2020). Low light and the persistence of stable stratification might together explain why phytoplankton communities are dominated by mixotrophic and motile forms under ice (Bertilsson et al., 2013; McKnight et al., 2000; Özkundakci et al., 2016; Pennak, 1968), which facilitates the acquisition of resources in a spatially or temporally heterogeneous environment. Mixotrophs can have a two-fold advantage in ice covered lakes-first, they are usually motile, hence, they can actively adjust their position within the water column and access to light, and second, they can supplement phototrophy with organic carbon sources (bacteria, DOM) under resource limitation (McKnight et al., 1993; Rue et al., 2019; Stoecker, 1998). By consuming and exuding organic material, heterotrophic organisms contribute to the internal loading of nutrient and carbon pools, subsidizing the under-ice food web (Fitch et al., 2018; Rue et al., 2019; Stockner & Porter, 1988; Thompson & Cotner, 2018, 2020).

Lower primary production under ice with snow cover has a range of implications for higher trophic levels. Zooplankton and zoobenthos have lower production rates at lower temperatures (Figure 2.3c; Shuter & Ing, 1997; Sutcliffe et al., 1981; Strayer, 2009), and may be supported primarily by mixotrophs (Perga et al., 2020) and microbial production (Bastviken et al., 2003). Zooplankton secondary or tertiary winter production peaks are restricted to early winter (Beaver et al., 2018; Grosbois et al., 2020). Most cladocerans and some copepods produce resting eggs and disappear from the water column, while other copepods are inactive through the winter and others stay active with the help of accumulated fat reserves (Schneider et al., 2017). Fat reserves can also be allocated to egg production in late winter immediately prior to the increasing light levels and the return of phytoplankton production (Schneider et al., 2017). The profundal zoobenthos has been shown to experience a mid-winter pro-



ductivity minimum (Dermott et al., 1977; Jónasson, 1975), which is attributed to reduced inputs of sedimenting pelagic production more than to low temperatures, since annual temperature variation in profundal sediments is usually low in stratified lakes. Since profundal organisms exhibit diverse feeding strategies, they are differentially affected by the reduced winter supply of sedimenting organic matter. Littoral zoobenthos show a larger seasonal drop in production compared to profundal zoobenthos, since they experience both a sharper reduction in temperature and a decrease in their food base due to suppression of benthic primary production (Jónasson, 1996).

Fish have a diversity of winter coping strategies to survive cold, dark, and limited prey conditions (Shuter et al., 2012). Winter-inactive fish experience dramatic energy savings from reduced swimming and foraging, and can survive winter with little to no feeding (Speers-Roesch et al., 2018). Winter-active fish possess the physiological capacity to sustain foraging when it is cold and dark (Blanchfield et al., 2009; Helland et al., 2011), although activity rates can decline in winter in some species (Watson et al., 2019). Variation in winter activity also exists among individuals of the same species (Auer et al., 2016). For example, smaller-bodied individuals are thought to be more susceptible to overwinter mortality (Byström et al., 2006; Hurst, 2007). The local environment also plays a role for fish activity and survival over winter. Widespread and persistent hypoxic and anoxic conditions in some snow and ice-covered lakes can result in a higher probability of fish-kills (Barica & Mathias, 1979; Greenbank, 1945). In other lakes, fish can behaviorally cope with and survive winter by moving to available habitats with more suitable temperature, oxygen, and light conditions (Blanchfield et al., 2009; Magnuson et al., 1985).

3.2. Clear Ice Cover With No Snow

Clear ice cover without any snow allows for increased transmittance of solar radiation in the water column. The extent of the differences between clear ice and snow + ice conditions will depend on the optical properties and thickness of the ice. Overall higher input of solar radiation under clear ice generates convective mixing, which leads to relatively deeper mixing and euphotic zones (Figures 2.1b, 2.2a, and 2.2c). These conditions allow for the formation of chlorophyll-*a* maxima in the euphotic zone (Figures 2.2f, 4 and 5), leading to higher primary production and phytoplankton biomass (Figure 2.3c). Higher resource availability will also stimulate secondary production (Figure 2.3c). While conditions in the upper water column will likely be oxygen-saturated, anoxia can develop in the bottom waters, exacerbating gradients of redox and solutes (Figure 2.2). Lakes that experience little or no snow may be expected to have either shorter or longer ice duration than lakes that remain snow-covered, depending on variation in ice-structure and thickness—a pattern that could be explored across climate gradients.

The absence of snow cover increases the transmittance of solar radiation to the water column (Figures 3 and 4), but the extent and intensity of the heat flux depends on the quality and thickness of the ice. Ice that has undergone freeze-thaw cycles, includes debris, snow, or gas bubbles will reflect or scatter more downwelling radiation and have lower transmission than clear, congelation ice, which can be extremely transparent to many wavelengths (Leppäranta, 2015; Warren, 2019). It is common for large lakes with long fetches to be snow free at many points during the winter due to blowing snow (Agbeti & Smon, 1995). Without the insulating effect of snow, ice thickness is more dependent on local climate forcing (Gould & Jeffries, 2005), potentially affecting light penetration. The overall lower albedo of ice compared to snow increases convective mixing through radiative gain (Figure 2.1b), decreasing water column stability with measured mixing rates of up to 7 mm s⁻¹ (Bouffard et al., 2016; Kirillin et al., 2012).

Radiative inputs through clear ice allow for higher primary production than when there is snow on the ice (Figure 2.3c). The primary production peak will extend deeper (Figures 2.2f, 4, and 5) due to higher light intensity and convective mixing. Under clear-ice conditions, phytoplankton biomass may be comparable to or exceed summer concentrations (Pasztaleniec & Lenard, 2008; Twiss et al., 2012; Vörös et al., 2009). Compared to winters when lake ice is covered with snow, non-motile algae are often abundant under clear ice and massive blooms of large diatoms have been frequently reported from large lakes experiencing such conditions (Beall et al., 2015; Bondarenko et al., 2006; Kerfoot et al., 2010). Moreover, clear ice itself can be a productive habitat for algae and bacteria (Bondarenko et al., 2012). While radiatively driven convective mixing can help non-motile algae to remain in suspension and increase their access to deeper nutrients (Katz et al., 2015; Kelley, 1997), it can also compromise their access to light if the mixing depth is too deep (Diehl et al., 2002a, 2000b). Small phytoplankton taxa with traits such as motility and mixotrophy appear to be successful under clear ice or ice and snow (Butts & Carrick, 2017; Callieri et al., 2006; Özkundakci et al., 2016). Small cell size allows for efficient



resource uptake, fast growth, and minimizes sinking rates, but these species generally have low grazing resistance (Chisholm, 1992; Litchman & Klausmeier, 2008).

Differences between surface and deeper water biogeochemistry will be accentuated under clear ice compared to ice with snow. Greater primary production and convective mixing would result in higher dissolved oxygen in the upper water column (Yang et al., 2017, 2020), while the hypolimnion may be more oxygen-depleted (Figures 2.2e and 4), possibly as a result of sinking autochthonous carbon produced during the winter, in addition to legacy or refractory sources of organic matter. Accordingly, we expect that there would be strong differences in net metabolism between the surface and deep waters. However, in more shallow lakes where the photic zone could extend to the bottom, these gradients would likely shift toward the sediment-water interface. The more oxic environment found under clear ice will enhance nitrification-with NH₄ oxidation being a substantial sink for oxygen where NH₄ is abundant (Figure 2.2l; Cavaliere & Baulch, 2019; Massé et al., 2019; Powers, Baulch, et al., 2017). As a consequence of nitrification and phytoplankton uptake, NH_4 concentrations may be lower within the euphotic zone (Figure 2.2m). Denitrifying bacteria may take advantage of accumulated organic matter and nitrate (NO_3) (Cavaliere & Baulch, 2018; Palacin-Lizarbe et al., 2018). Increased nitrification rates and nitrate accumulation could also affect phytoplankton community structure. While some phytoplankton preferentially take up NH,competing with nitrifiers for that substrate-diatoms have preferential uptake of NO₃ (Lomas & Glibert, 1999). Uptake and assimilation rates of NO_3 are high at low temperatures (Glibert et al., 2016); therefore, diatom growth could be enhanced if light conditions are favorable.

Given the higher rates of biogenic oxygen production in clear ice lakes, mineralization of organic matter may be more rapid, affecting nutrient availability, DOM lability (Rue et al., 2019), and GHG production (Jansen et al., 2019) during the winter and into spring (Tõnno et al., 2005). Subsequent mineralization of both settled particulate organic matter (POM) and DOM in bottom anoxic waters during winter may regenerate nutrient pools, as well as replenish reduced nitrogen species (Figure 2.2m) and phosphorus (Figure 2.2k; Cavaliere & Baulch, 2020; Rue et al., 2019). The flux of P to the upper water column is also influenced by redox conditions at the sediment-water interface (Nürnberg et al., 2013), so if the convectively mixed layer extends to lake sediments and keeps them oxygenated, sediment P fluxes may remain low (Markelov et al., 2019).

Phytoplankton productivity under clear ice may stimulate production at other trophic levels, even if the strong temperature dependence of zooplankton, zoobenthos and fish production keeps their overall growth rates low (Figure 2.3c; Bramm et al., 2009; Bertilsson et al., 2013; Sommer et al., 2012). For example, some copepods and, less commonly, cladocerans are abundant during winter and have reproductive strategies that enable them to rapidly respond to increasing phytoplankton production (Beaver et al., 2018). Active zooplankton grazing may enhance nutrient recycling, and also interact with phytoplankton by reducing their biomass, and changing their community structure, since taxa such as small diatoms and cryptomonads are preferentially grazed (Hrycik & Stockwell, 2020). Senescent phytoplankton biomass can stimulate production of microbes (Bižić-Ionescu et al., 2014) and profundal zoobenthos (Jónasson, 1975). Fish that remain active during the winter may experience increased predation rates or foraging efficiency because of higher light levels (Fullerton et al., 2000; Garvey et al., 2004; Varpe et al., 2015). Ice without snow could therefore support greater fish biomass compared to ice with snow conditions. Overall, increased light leads to greater biomass at the base of the food web, but further research is needed to understand how temperature regulated metabolic rates or shifts in phytoplankton community structure might restrict biomass to be transferred to the higher trophic levels (Figure 2.3a; Vad et al., 2020).

3.3. Loss of Winter Lake Ice Cover

In ice-free winters (Figure 2.1c), lakes experience constant interaction with the atmosphere and potentially greater exogenous inputs from the landscape due to precipitation falling mainly as rain (Figure 2.3b). Under these conditions, most of the incoming solar radiation is transmitted and absorbed by the water column, increasing z_{eu} (Figures 2.1c and 2.2d), which could contribute to higher primary production. However, this increase could be offset by greater wind-driven mixing (Figure 2.1c), which may deepen the mixing zone (z_m , Figure 2.2a) and could limit light availability depending on z_m/z_{eu} (Figure 2.3b). The increased mixing would also lead to a more uniform distribution of gases, ions, and nutrients throughout the water column (Figures 2.2 and 2.3a). Understanding how the frequency of ice-free winters shifts with regional and global climate change represents an important future research goal.



The loss of winter ice cover has profound implications for lake ecosystems through exposure to the atmosphere and wind energy. In general, wind-driven turbulence is expected to lead to greater mixing (Figures 2.1c and 2.2a), and the consequences of this mixing for thermal regimes will depend on regional and local factors affecting the energy budget (Box 3; Woolway & Merchant, 2019). Shallower lakes may mix continuously, but in deeper lakes, full mixing of the water column may be either substantially delayed (Box 3), or lakes may not fully turnover as a result of insufficient winter cooling and/or high solute concentrations (Box 3; Woolway et al., 2020). If full turnover during winter is completely halted, redistribution of gases and nutrients will be limited, resulting in persistent anoxia in bottom waters and resource-limitation in the upper water column (Lau et al., 2020; Yankova et al., 2017). Mixing will reduce the strength of vertical physio-chemical gradients (Figure 2.3a) and increase the supply of dissolved oxygen to the bottom waters (Figure 2.2e). Coupled with higher lakewater temperatures (Figures 2.2b and 5; Flicker et al., 2017), higher oxygen availability may lead to reduced carbon burial efficiency, and enhanced mineralization rates of more readily degradable organic matter, potentially increasing refractive carbon pools (Gudasz et al., 2010, 2015; Sobek et al., 2009). A well-mixed water column also has the potential to redistribute sediments through the water column, with implications for turbidity, nutrients, and trace metals (Evans, 1994). Enhanced resuspension of sediments could also affect community structure and quality of prey by lifting plankton resting cysts from the sediments, inoculating the water column (e.g., akinetes of cyanobacteria and zooplankton eggs; Cirés et al., 2013; Özkundakci et al., 2016). Loss of ice and increased exposure to the atmosphere could also contribute to increased evaporation, gas exchange, and atmospheric deposition (Arp et al., 2015; Xiao et al., 2018).

Increased connectivity of lakes with their watersheds during ice-free winters (Figure 2.1c) will likely result in a change in the timing and/or magnitude of exogenous inputs of carbon and nutrients. The amount and composition of the input will depend on the local climate, precipitation patterns, hydrology, watershed size, and the type of land cover surrounding the lakes (Casson et al., 2019; Solomon et al., 2015). Similar terrestrial loading may also occur in ice-covered lakes if soils do not freeze during the winter (Melack et al., 2020). We expect that delivery of terrestrial inputs into ice-free lakes during winter will be primarily mediated by the frequency of rain and freeze-thaw events. While there is a scarcity of evidence of lake responses to open-water winters, warmer winters with more frequent rain events are associated with higher riverine dissolved organic carbon and nutrient exports (Huntington et al., 2016; Rattan et al., 2017). Increased runoff and loading of sediments, dissolved nutrients, and organic matter during ice-free winters could alter redox potential (Jansen et al., 2019) and exacerbate increasing turbidity, nutrient enrichment and/or browning of lakes (Cade-Menun et al., 2013; Box 3). Thus, net lake metabolism in winter under open water conditions may be more strongly influenced by landscape features. While decreasing water clarity with associated effects for stratification patterns, lake metabolism, and food webs has been documented during the summer (Hayden et al., 2019; Leech et al., 2018; Staehr et al., 2010), how enhanced connectivity to the landscape during winter will affect lake responses requires further exploration.

It is tempting to conclude that lower albedo and increased radiative inputs (Figure 2.2d) in winter ice-free lakes would result in the highest rates of primary production along the LICC. However, increased mixing will likely reduce residence time of phytoplankton within the euphotic zone, leading to light limitation and lower photosynthetic activity (higher z_m/z_{en} , Figures 2.2a, 2.2c, 2.3b, 2.3c, and 3; Diehl et al., 2002b; Vincent, 1983). Moreover, increased mixing might also result in higher turbidity (Beall et al., 2015), reducing light availability and depth of the photic zone (Box 3). In Lake Erie, a shift from near complete ice coverage to largely open-water winter conditions led to a > 70% decline in mid-February chlorophyll-a (Beall et al., 2015). These stark differences in chlorophyll-a under differing winter conditions persisted into spring (Beall et al., 2015). Similarly, spring diatom production in Tiefer See, Germany, was low after winters with low ice cover and prolonged mixing periods (Kienel et al., 2017). During ice-free winters, the phytoplankton community may shift toward one dominated by non-motile forms capable of withstanding fluctuating light in the mixed water column (e.g., diatoms and some cyanobacteria; Litchman, 1998; Shatwell et al., 2008); however, the distribution of their biomass will depend on the frequency, intensity, and depth of the mixing (Box 3; Huisman et al., 1999). Consequently, lake metabolic balance will also depend on mixing dynamics and factors controlling the depth of the euphotic zone, such as lake morphometry, water clarity, and rates of watershed loading (Obrador et al., 2014; Olson et al., 2020; Kelly et al., 2018).

Secondary production of invertebrates is expected to increase during winter under ice-free conditions as a result of the relatively higher temperatures, especially in the upper water column, and resulting increased food availa-



bility (Figure 2.3c); while primary production may be lower compared to clear ice conditions, it will be higher relative to snow-covered winters (Figure 2.3c). Massive swarms of reproducing cladocerans have been observed at water temperatures below 10 °C in Lake Mead (Beaver et al., 2018). Similarly, cladocerans can be abundant in Lake Washington, which remains ice-free during winters, with water temperatures typically ranging between 6.7 and 9.4 °C (Hampton et al., 2006). This suggests that an increase in water temperature by even a few degrees could lead to substantial changes in zooplankton production. Profundal benthic organisms could also benefit from ice loss due to a combination of increased temperatures and increased supply of food to filter feeders and surface-dwelling detritivores by deeper mixing (Rowe et al., 2017; Vanderploeg et al., 2010). Similarly, littoral zoobenthos production will likely be stimulated by littoral primary production that is expected to increase in response to higher light availability. Although rates of fish foraging and production could increase (Figure 2.3c; Varpe et al., 2015), foraging success and fish survival may decline if visibility is reduced as a result of wind driven wave action and shoreline erosion, resuspension of sediments, or increased turbidity (Hedström et al., 2016). Warmer spring temperatures might inhibit successful foraging of cold-water fish in productive, littoral zones (Caldwell, Chandra, Feher, et al., 2020; Guzzo & Blanchfield, 2017; Guzzo et al., 2017; Jansen & Hesslein, 2004) and decrease juvenile abundance (Farmer et al., 2015). While these examples show responses to earlier spring warming, not the lack of ice per se, these studies stress the importance of considering how fish behavior and foraging success during winters with no ice carries over, affecting growth and reproduction in subsequent seasons.

Considering the substantial impact of ice cover on lake ecosystems, a shift toward year-round open-water conditions will affect habitat availability and likely alter the structure of biological communities. Differential responses among longer-lived organisms to cold, harsh conditions during winter contribute to stabilization of food webs and maintenance of diversity (Humphries et al., 2017; McMeans et al., 2015). For example, coexistence among fish with different thermal preferences is likely mediated by seasonal differences in habitat use, behavior, and fitness (Farmer et al., 2015; McMeans et al., 2020). Therefore, the loss of cold winters and resulting seasonal homogenization can affect population dynamics (e.g., lower winter mortality), resource use, and species interactions—having cascading effects on ecosystem function. Similarly, we could expect that homogenization of vertical gradients under full winter mixing may affect the structure of microbial communities taking advantage of spatial variability, for example in redox conditions (Figure 2.3a; Schütte et al., 2016). Finally, Timoshkin (2001) has called winter ice the "second bottom," referring to the idea that ice creates a stable surface where growth can occur in the photic zone. Therefore, the disappearance of ice cover itself represents a loss of habitat for cold-adapted microorganisms living attached to (e.g., diatoms; Bondarenko et al., 2006), inside (e.g., bacteria; Santibáñez et al., 2019), or on top of the ice cover (e.g., flagellates and ciliates; Felip et al., 1999). Taken together, we expect that the loss of ice cover and cold conditions during winter will have consequences for coexistence of species and annual patterns of biodiversity.

4. Implications of Declining Ice Duration

Our conceptual framework highlights the wide range of lake responses to ongoing climate-driven changes in winter lake ice cover. The shorter ice duration observed throughout the world has manifested as both later freeze and earlier ice breakup dates, strongly affected by temperatures in the months preceding these events (Magnuson, 2000; Shatwell et al., 2019). Such temporal connections serve as a reminder of the need to "close the loop" and understand the role of winter in the aquatic annual cycle (Salonen et al., 2009). For example, earlier ice-out will affect available light levels in the water, even though both day length and angle of incidence are lower earlier in the year (Preston et al., 2016), and autocorrelation of water temperatures means that warmer winter temperatures will often beget warmer spring temperatures (Box 3; Adrian et al., 1999; Caldwell, Chandra, Albright, et al., 2020). The accumulation, depletion or transformation of nutrient and carbon pools under ice (e.g., Cavaliere & Baulch, 2020; Powers, Labou, et al., 2017; Rue et al., 2019) will set the stage for early spring primary producers. For example, high concentrations of chlorophyll-*a* during winter were associated with low concentration of chlorophyll-*a* during the following summer, possibly due to the reduction of the pelagic nutrient pool as a result of higher winter productivity (North et al., 2014).

In years with shorter winters, the phenology of important events can be altered. For example, lakes that typically receive large influxes of spring melt water will likely receive that water, often rich in nutrients, earlier in the year (Sadro et al., 2018). Also, linkages to the landscape may persist for a longer portion of the year, delivering



allochthonous inputs (Haei et al., 2013) and potentially releasing GHGs (Denfeld et al., 2018) more consistently than as a pulse (Lake, 2000). The degree to which increased landscape connectivity influences lake processes both during winter and across seasons will depend upon lake (e.g., residence time), local (e.g., land use and watershed:lake area), and regional (e.g., climate) attributes. These interacting processes may also increase or buffer susceptibility of lakes to changes in trophic status, as well as the effects of eutrophication and humification on nutrient cycles and energy transfer efficiency of food webs therein (Karpowicz et al., 2020). However, the impact on biogeochemical processes and biological communities through the annual cycle (Missaghi et al., 2017; Orihel et al., 2017) will also largely depend on how shorter winters affect the duration and intensity of spring mixing given their connection with hypolimnetic temperatures and oxygen depletion (Dugan, 2021a; North et al., 2014).

As more lakes move toward ice-free winters, radical changes in ecosystem processes will undoubtedly be observed for many lakes (Figures 2 and 5). The complexity of nutrient influx, accumulation, and processing under changing winter conditions has uncertain implications for phytoplankton composition and biomass both under ice and at spring melt, as highly varied relationships of nutrients and ice duration have now been observed across systems. In turn, conditions that favor either high winter algal biomass or high spring algal biomass are expected to fuel zooplankton growth that can be transferred to fish and benthos, particularly under warming conditions. However, shifts in winter phenology may also result in mismatches among trophic levels (Thackeray et al., 2010; Weyhenmeyer et al., 2008) and systems susceptible to harmful algal blooms could experience further water quality problems associated with enhanced growth of cyanobacteria (Anneville et al., 2015; Ho & Michalak, 2019). "Winners" and "losers" will emerge across all trophic levels (Dornelas et al., 2019) given that annual patterns of biodiversity are, in part, maintained by the specialized niches created by contrasts between winter and summer (McMeans et al., 2020; Shatwell et al., 2008). Loss of winter habitat will reduce diversity and abundance of cold stenotherms or organisms otherwise specialized for growing under ice, such as those attached to ice (e.g., Bondarenko et al., 2006). This phenomenon is already evident among fish, such that major changes in winter duration would be expected to alter community composition throughout the year (McMeans et al., 2020; Shuter et al., 2016), with the potential for cascading ecological effects.

5. Conclusions

Global change is driving ice-covered lakes toward not only warmer annual average temperatures, but also reduced, intermittent or no winter ice cover. Concurrently, these lakes are experiencing shifts in the timing and intensity of precipitation, as well as the proportion arriving to lakes as rain or snow. For lakes that experience ice cover, it is well understood that ice duration, thickness, clarity, and snow cover together influence physical, chemical, and biological processes. The open questions concern the exact nature of these complex, interacting effects in any lake ecosystem and the repercussions of winter changes throughout the year. We expect seasonal linkages resulting from changing winter conditions to vary within individual lakes, affected by an assortment of factors including the timing and magnitude of inflowing melt waters, lake size and morphometry, and landscape variation from the continental to the local scale. At this stage in the development of winter limnology, we see an essential need to better articulate the elements of winter that appear generalizable enough to provide a useful research paradigm, both to advance fundamental knowledge and to anticipate future ecosystem changes. Accordingly, we propose the LICC as an organizing framework for future research. Conceptual frameworks have been vital for advancing our understanding of ecosystem processes and how they change across scales (Raymond et al., 2016; Seastedt et al., 2004; Vannote et al., 1980; Ward & Stanford, 1982). As such, the LICC contributes a new approach for characterizing lakes under a continuum of winter conditions that mediate energy fluxes into lakes in a landscape context, which can be used as a tool to test our knowledge of mechanistic linkages and ecosystem responses to globally changing winters. The pace of worldwide lake warming, decreasing ice duration, and altered precipitation patterns lends imperative to the establishment of this framework for addressing key questions in winter limnology and advancing discovery while ice still dominates the world's lakes.

Data Availability Statement

Lake Mendota monitoring data and figure code are available in the Zenodo open-access data repository supported by the Conseil Européen pour la Recherche Nucléaire (https://doi.org/10.5281/zenodo.4812868). Lake Stechlin monitoring data, figure code, and photo are available from the Freshwater Research and Environmental Database



Acknowledgments

The inception of ideas for this manuscript was the outcome of discussions among all authors at the AGU Chapman Conference "Winter Limnology in a Changing World," and the authors are grateful to the staff of American Geophysical Union, and those of Flathead Lake Biological Station where the meeting was held. This was a collaborative group with many meetings, subgroups, and widely distributed effort with people involved in all aspects of iteratively developing the manuscript. The first co-lead author group (E. Cavaliere, I. B. Fournier, V. Hazuková, G. P. Rue, and S. Sadro) is alphabetically ordered due to equal contributions in leadership and effort over the duration of the project. The second author group (S. A. Berger, J. B. Cotner, H. A Dugan, S. E. Hampton, N. R. Lottig, B. C. McMeans, T. Ozersky, S. M. Powers, and M. Rautio) is alphabetically ordered due to equal contribution. All authors contributed to idea development and original text, and provided substantive feedback. E. Cavaliere, H. A. Dugan, I. B. Fournier, and S. Sadro led subteams to review the literature. Figure development was led by S. A. Berger, E. Cavaliere, H. A. Dugan, I. B. Fournier, V. Hazuková, C. M. O'Reilly, S. M. Powers, and G. P. Rue. C. M. O'Reilly coordinated the project. The authors thank Shawn Devlin for input and discussions. Lake Mendota observations were supported by the US National Science Foundation #DEB-1440297 and #DEB-1856224. Lake Stechlin data were collected as part of the Monitoring Program supported by the Department of Experimental Limnology at IGB Stechlin, Germany. Lake Simoncouche data were collected by the infrastructure provided by the Canada Foundation for Innovation grant #35253 and with support from the Canada Research Chairs Program.

(FRED) operated by the Leibniz Institute of Freshwater Ecology and Inland Fisheries (https://doi.org/10.18728/ igb-fred-576.0). The Lake Simoncouche data monitoring data are available from the Centre for Northern Studies data repository Nordicana D (https://doi.org/10.5885/45717CE-5A9BFDF1C9064936).

References

- Adrian, R., Walz, N., Hintze, T., Hoeg, S., & Rusche, R. (1999). Effects of ice duration on plankton succession during spring in a shallow polymictic lake. *Freshwater Biology*, 41(3), 621–634. https://doi.org/10.1046/j.1365-2427.1999.00411.x
- Agbeti, M. D., & Smol, J. P. (1995). Winter limnology: A comparison of physical, chemical and biological characteristics in two temperate lakes during ice cover. *Hydrobiologia*, 304(3), 221–234. https://doi.org/10.1007/bf02329316
- Anneville, O., Domaizon, I., Kerimoglu, O., Rimet, F., & Jacquet, S. (2015). Blue-green algae in a "Greenhouse Century"? New insights from field data on climate change impacts on cyanobacteria abundance. *Ecosystems*, 18(3), 441–458. https://doi.org/10.1007/s10021-014-9837-6
- Arp, C. D., Jones, B. M., Liljedahl, A. K., Hinkel, K. M., & Welker, J. A. (2015). Depth, ice thickness, and ice-out timing cause divergent hydrologic responses among Arctic lakes. *Water Resources Research*, 51(12), 9379–9401. https://doi.org/10.1002/2015wr017362
- Auer, S. K., Salin, K., Anderson, G. J., & Metcalfe, N. B. (2016). Flexibility in metabolic rate and activity level determines individual variation in overwinter performance. *Oecologia*, *182*(3), 703–712. https://doi.org/10.1007/s00442-016-3697-z
- Barica, J., & Mathias, J. A. (1979). Oxygen depletion and winterkill risk in small Prairie lakes under extended ice cover. Journal of the Fisheries Research Board of Canada, 36(8), 980–986. https://doi.org/10.1139/t79-136
- Bastviken, D., Ejlertsson, J., Sundh, I., & Tranvik, L. (2003). Methane as a source of carbon and energy for lake pelagic food webs. *Ecology*, 84(4), 969–981. https://doi.org/10.1890/0012-9658(2003)084[0969:MAASOC]2.0.CO;2
- Beall, B. F. N., Twiss, M. R., Smith, D. E., Oyserman, B. O., Rozmarynowycz, M. J., Binding, C. E., et al. (2015). Ice cover extent drives phytoplankton and bacterial community structure in a large north-temperate lake: Implications for a warming climate. *Environmental Microbiology*, 18(6), 1704–1719. https://doi.org/10.1111/1462-2920.12819
- Beaver, J. R., Kirsch, J. E., Tausz, C. E., Samples, E. E., Renicker, T. R., Scotese, K. C., et al. (2018). Long-term trends in seasonal plankton dynamics in Lake Mead (Nevada-Arizona, USA) and implications for climate change. *Hydrobiologia*, 822(1), 85–109. https://doi.org/10.1007/ s10750-018-3638-4
- Bégin, P. N., Rautio, M., Tanabe, Y., Uchida, M., Culley, A. I., & Vincent, W. F. (2021). The littoral zone of polar lakes: Inshore–offshore contrasts in an ice-covered High Arctic lake. Arctic Science, 7(1), 158–181. https://doi.org/10.1139/as-2020-0026
- Benson, B. J., Magnuson, J. J., Jensen, O. P., Card, V. M., Hodgkins, G., Korhonen, J., et al. (2012). Extreme events, trends, and variability in Northern Hemisphere lake-ice phenology (1855–2005). *Climatic Change*, 112(2), 299–323. https://doi.org/10.1007/s10584-011-0212-8
- Berger, S. A., Cavaliere, E., Fournier, I., & Hazukova, V. (2021). Lake Stechlin winter and spring water column profiles addressing cross-seasonal linkages in cold and warm years 2012 and 2020 (The Lake Ice Continuum Concept). *IGB Freshwater Research and Environmental Database*. https://doi.org/10.18728/igb-fred-576.0
- Bertilsson, S., Burgin, A., Carey, C. C., Fey, S. B., Grossart, H., Grubisic, L. M., et al. (2013). The under-ice microbiome of seasonally frozen lakes. *Limnology & Oceanography*, 58(6), 1998–2012. https://doi.org/10.4319/lo.2013.58.6.1998
- Bižić-Ionescu, M., Amann, R., & Grossart, H. P. (2014). Massive regime shifts and high activity of heterotrophic bacteria in an ice-covered lake. *PLoS One*, 9(11), 117. https://doi.org/10.1371/journal.pone.0113611
- Blanchfield, P. J., Tate, L. S., Plumb, J. M., Acolas, M., & Beaty, K. G. (2009). Seasonal habitat selection by lake trout (*Salvelinus namay-cush*) in a small Canadian Shield lake: Constraints imposed by winter conditions. *Aquatic Ecology*, 43(3), 777–787. https://doi.org/10.1007/s10452-009-9266-3
- Bondarenko, N. A., Belykh, O. I., Golobokova, L. P., Artemyeva, O. V., Logacheva, N. F., Tikhonova, I. V., et al. (2012). Stratified distribution of nutrients and extremophile biota within freshwater ice covering the surface of Lake Baikal. *Journal of Microbiology*, 50(1), 8–16. https:// doi.org/10.1007/s12275-012-1251-1
- Bondarenko, N. A., Timoshkin, O. A., Röpstorf, P., & Melnik, N. G. (2006). The under-ice and bottom periods in the life cycle of Aulacoseira baicalensis (K. Meyer) Simonsen, a principal Lake Baikal alga. Hydrobiologia, 568(Suppl. 1), 107–109. https://doi.org/10.1007/ s10750-006-0325-7
- Bouffard, D., Zdorovennov, R. E., Zdorovennova, G. E., Pasche, N., Wüest, A., & Terzhevik, A. Y. (2016). Ice-covered Lake Onega: Effects of radiation on convection and internal waves. *Hydrobiologia*, 780, 21–36. https://doi.org/10.1007/s10750-016-2915-3
- Bramm, M. E., Lassen, M. K., Liboriussen, L., Richardson, K., Ventura, M., & Jeppesen, E. (2009). The role of light for fish-zooplankton-phytoplankton interactions during winter in shallow lakes - A climate change perspective. *Freshwater Biology*, 54(5), 1093–1109. https://doi. org/10.1111/j.1365-2427.2008.02156.x
- Brentrup, J. A., Richardson, D. C., Carey, C. C., Ward, N. K., Bruesewitz, D. A., & Weathers, K. C. (2020). Under-ice respiration rates shift the annual carbon cycle in the mixed layer of an oligotrophic lake from autotrophy to heterotrophy. *Inland Waters*, 11, 1–123. https://doi.org/10. 1080/20442041.2020.1805261
- Bruesewitz, D. A., Carey, C. C., Richardson, D. C., & Weathers, K. C. (2015). Under-ice thermal stratification dynamics of a large, deep lake revealed by high-frequency data. *Limnology & Oceanography*, 60(2), 347–359. https://doi.org/10.1002/lno.10014
- Butts, E., & Carrick, H. J. (2017). Phytoplankton seasonality along a trophic gradient of temperate lakes: Convergence in taxonomic composition during winter ice-cover. Northeastern Naturalist, 24(sp7), B167–B187. https://doi.org/10.1656/045.024.s719
- Byström, P., Andersson, J., Kiessling, A., & Eriksson, L. (2006). Size and temperature dependent foraging capacities and metabolism: Consequences for winter starvation mortality in fish. *Oikos*, 115(1), 43–52. https://doi.org/10.1111/j.2006.0030-1299.15014.x
- Cade-Menun, B. J., Bell, G., Baker-Ismail, S., Fouli, Y., Hodder, K., McMartin, D. W., et al. (2013). Nutrient loss from Saskatchewan cropland and pasture in spring snowmelt runoff. *Canadian Journal of Soil Science*, 93(4), 445–458. https://doi.org/10.4141/cjss2012-042
- Caldwell, T. J., Chandra, S., Albright, T. P., Harpold, A. A., Dilts, T. E., Greenberg, J. A., et al. (2020). Drivers and projections of ice phenology in mountain lakes in the western United States. *Limnology & Oceanography*, 66, 995–1008. https://doi.org/10.1002/lno.11656
- Caldwell, T. J., Chandra, S., Feher, K., Simmons, J. B., & Hogan, Z. (2020). Ecosystem response to earlier ice break-up date: Climate-driven changes to water temperature, lake-habitat-specific production, and trout habitat and resource use. *Global Change Biology*, 26(10), 5475–5491. https://doi.org/10.1111/gcb.15258
- Callieri, C., Caravati, E., Morabito, G., & Oggioni, A. (2006). The unicellular freshwater cyanobacterium *Synechococcus* and mixotrophic flagellates: Evidence for a functional association in an oligotrophic, subalpine lake. *Freshwater Biology*, *51*(2), 263–273. https://doi.org/10.1111/j.1365-2427.2005.01489.x



- Casson, N. J., Wilson, H. F., & Higgins, S. M. (2019). Hydrological and seasonal controls of phosphorus in Northern Great Plains agricultural streams. *Journal of Environmental Quality*, 48(4), 978–987. https://doi.org/10.2134/jeq2018.07.0281
- Castendyk, D. N., Obryk, M. K., Leidman, S. Z., Gooseff, M., & Hawes, I. (2016). Lake Vanda: A sentinel for climate change in the McMurdo Sound Region of Antarctica. *Global and Planetary Change*, 144, 213–227. https://doi.org/10.1016/j.gloplacha.2016.06.007
- Catalan, J. (1992). Evolution of dissolved and particulate matter during the ice-covered period in a deep, high-mountain lake. *Canadian Journal of Fisheries and Aquatic Sciences*, 49(5), 945–955. https://doi.org/10.1139/f92-105
- Cavaliere, E., & Baulch, H. M. (2018). Denitrification under lake ice. *Biogeochemistry*, 137(3), 285–295. https://doi.org/10.1007/s10533-018-0419-0
- Cavaliere, E., & Baulch, H. M. (2019). Winter nitrification in ice-covered lakes. *PLoS One*, 14(11), e0224864. https://doi.org/10.1371/journal. pone.0224864
- Cavaliere, E., & Baulch, H. M. (2020). Winter in two phases: Long-term study of a shallow reservoir in winter. *Limnology & Oceanography*, 66, 1335–1352. https://doi.org/10.1002/lno.11687
- Chisholm, S. W. (1992). Phytoplankton Size. In P. G. Falkowski, & A. D. Woodhead (Eds.), Primary productivity and biogeochemical cycles in the sea (pp. 213–237). https://doi.org/10.1007/978-1-4899-0762-2_12
- Cirés, S., Wörmer, L., Agha, R., & Quesada, A. (2013). Overwintering populations of Anabaena, Aphanizomenon and Microcystis as potential inocula for summer blooms. Journal of Plankton Research, 35(6), 1254–1266. https://doi.org/10.1093/plankt/fbt081
- Davison, I. R. (1991). Environmental effects on algal photosynthesis: Temperature. Journal of Phycology, 27(1), 2–8. https://doi. org/10.1111/j.0022-3646.1991.00002.x
- Denfeld, B. A., Baulch, H. M., del Giorgio, P. A., Hampton, S. E., & Karlsson, J. (2018). A synthesis of carbon dioxide and methane dynamics during the ice-covered period of northern lakes. *Limnology & Oceanography Letters*, 3(3), 117–131. https://doi.org/10.1002/lol2.10079
- Dermott, R. M., Kalff, J., Leggett, W. C., & Spence, J. (1977). Production of Chironomus, Procladius, and Chaoborus at different levels of phytoplankton biomass in Lake Memphremagog, Quebec–Vermont. Journal of the Fisheries Research Board of Canada, 34(11), 2001–2007. https://doi.org/10.1139/f77-268
- Diehl, S., Berger, S., Ptacnik, R., & Wild, A. (2002a). Phytoplankton, light, and nutrients in a gradient of mixing depths: Theory. *Ecology*, 83(2), 386–398. https://doi.org/10.1890/0012-9658(2002)083[0386:PLANIA]2.0.CO;2
- Diehl, S., Berger, S., Ptacnik, R., & Wild, A. (2002b). Phytoplankton, light, and nutrients in a gradient of mixing depths: Field experiments. *Ecology*, 83(2), 399–411. https://doi.org/10.1890/0012-9658(2002)083[0399:PLANIA]2.0.CO;2
- Dillon, M. L., Hawes, I., Jungblut, A. D., Mackey, T. J., Eisen, J. A., Doran, P. T., & Sumner, D. Y. (2020). Environmental control on the distribution of metabolic strategies of benthic microbial mats in Lake Fryxell, Antarctica. *PLoS One*, 15(4), e0231053. https://doi.org/10.1371/ journal.pone.0231053
- Dornelas, M., Gotelli, N. J., Shimadzu, H., Moyes, F., Magurran, A. E., & McGill, B. J. (2019). A balance of winners and losers in the Anthropocene. *Ecology Letters*, 22(5), 847–854. https://doi.org/10.1111/ele.13242
- Dugan, H. A. (2021a). A Comparison of ecological memory of lake ice-off in eight north-temperate lakes. Journal of Geophysical Research: Biogeosciences, 126(6). https://doi.org/10.1029/2020jg006232
- Dugan, H. A. (2021b). Lake Mendota: Under-ice light and water column profiles 2018 and 2019 (The Lake Ice Continuum Concept). Zenodo. https://doi.org/10.5281/zenodo.4812868
- Dumanski, S., Pomeroy, J. W., & Westbrook, C. J. (2015). Hydrological regime changes in a Canadian Prairie basin. Hydrological Processes, 29(18), 3893–3904. https://doi.org/10.1002/hyp.10567
- Evans, R. D. (1994). Empirical evidence of the importance of sediment resuspension in lakes. *Hydrobiologia*, 284(1), 5–12. https://doi.org/10.1007/bf00005727
- Falkowski, P. G., & Raven, J. A. (2007). Aquatic photosynthesis. Princeton University Press. https://doi.org/10.1515/9781400849727
- Farmer, T. M., Marschall, E. A., Dabrowski, K., & Ludsin, S. A. (2015). Short winters threaten temperate fish populations. Nature Communications, 6(1). https://doi.org/10.1038/ncomms8724
- Felip, M., Camarero, L., & Catalan, J. (1999). Temporal changes of microbial assemblages in the ice and snow cover of a high mountain lake. Limnology & Oceanography, 44(4), 973–987. https://doi.org/10.4319/10.1999.44.4.0973
- Fernandez, I., Birkel, S., Schmitt, C., Simonson, J., Lyon, B., Pershing, A., et al. (2020). Maine's climate future 2020 update. University of Maine. https://doi.org/10.13140/RG.2.2.24401.07521
- Ficker, H., Luger, M., & Gassner, H. (2017). From dimictic to monomictic: Empirical evidence of thermal regime transitions in three deep alpine lakes in Austria induced by climate change. *Freshwater Biology*, 62, 1335–1345. https://doi.org/10.1111/fwb.12946
- Filazzola, A., Blagrave, K., Imrit, M. A., & Sharma, S. (2020). Climate change drives increases in extreme events for lake ice in the Northern Hemisphere. *Geophysical Research Letters*, 47(18). https://doi.org/10.1029/2020gl089608
- Fitch, A., Orland, C., Willer, D., Emilson, E. J., & Tanentzap, A. J. (2018). Feasting on terrestrial organic matter: Dining in a dark lake changes microbial decomposition. *Global Change Biology*, 24(11), 5110–5122. https://doi.org/10.1111/gcb.14391
- Fullerton, A. H., Garvey, J. E., Wright, R. A., & Stein, R. A. (2000). Overwinter growth and survival of largemouth bass: Interactions among size, food, origin, and winter severity. *Transactions of the American Fisheries Society*, 129(1), 1–12. https://doi.org/10.1577/1548-8659(2000)1292 .0.co;210.1577/1548-8659(2000)129<0001:ogasol>2.0.co;2
- Garvey, J. E., Ostrand, K. G., & Wahl, D. H. (2004). Energetics, predation, and ratio affect size-dependent growth and mortality of fish during winter. *Ecology*, 85(10), 2860–2871. https://doi.org/10.1890/03-0329
- Glibert, P. M., Wilkerson, F. P., Dugdale, R. C., Raven, J. A., Dupont, C. L., Leavitt, P. R., et al. (2016). Pluses and minuses of ammonium and nitrate uptake and assimilation by phytoplankton and implications for productivity and community composition, with emphasis on nitrogen-enriched conditions. *Limnology & Oceanography*, 61(1), 165–197. https://doi.org/10.1002/ino.10203
- Gonsior, M., Schmitt-Kopplin, P., & Bastviken, D. (2013). Depth-dependent molecular composition and photo-reactivity of dissolved organic matter in a boreal lake under winter and summer conditions. *Biogeosciences*, 10(11), 6945–6956. https://doi.org/10.5194/bg-10-6945-2013
- Gould, M., & Jeffries, M. (2005). Temperature variations in lake ice in central Alaska, USA. Annals of Glaciology, 40, 89–94. https://doi.org/10.3189/172756405781813825
- Granados, I., Toro, M., Giralt, S., Camacho, A., & Montes, C. (2020). Water column changes under ice during different winters in a mid-latitude Mediterranean high mountain lake. Aquatic Sciences, 82(2). https://doi.org/10.1007/s00027-020-0699-z
- Greenbank, J. (1945). Limnological conditions in ice-covered lakes, especially as related to winter-kill of fish. *Ecological Monographs*, 15(4), 343–392. https://doi.org/10.2307/1948427
- Grosbois, G., Mariash, H., Schneider, T., & Rautio, M. (2017). Under-ice availability of phytoplankton lipids is key to freshwater zooplankton winter survival. Scientific Reports, 7(1). https://doi.org/10.1038/s41598-017-10956-0



- Grosbois, G., Vachon, D., Del Giorgio, P. A., & Rautio, M. (2020). Efficiency of crustacean zooplankton in transferring allochthonous carbon in a boreal lake. *Ecology*, *101*(6). https://doi.org/10.1002/ecy.3013
- Gudasz, C., Bastviken, D., Steger, K., Premke, K., Sobek, S., & Tranvik, L. J. (2010). Temperature-controlled organic carbon mineralization in lake sediments. *Nature*, 466(7305), 478–481. https://doi.org/10.1038/nature09186
- Gudasz, C., Sobek, S., Bastviken, D., Koehler, B., & Tranvik, L. J. (2015). Temperature sensitivity of organic carbon mineralization in contrasting lake sediments. *Journal of Geophysical Research: Biogeosciences*, 120(7), 1215–1225. https://doi.org/10.1002/2015jg002928
- Guzzo, M. M., & Blanchfield, P. J. (2017). Climate change alters the quantity and phenology of habitat for lake trout (Salvelinus alpinus). Canadian Journal of Fisheries and Aquatic Sciences, 74, 871–884. https://doi.org/10.1139/cjfas-2016-0190
- Guzzo, M. M., Blanchfield, P. J., & Rennie, M. D. (2017). Behavioral responses to annual temperature variation alter the dominant energy pathway, growth, and condition of a cold-water predator. *Proceeding of the National Academy of Sciences USA*, 114, 9912–9917. https://doi. org/10.1073/pnas.1702584114
- Haei, M., Öquist, M. G., Kreyling, J., Ilstedt, U., & Laudon, H. (2013). Winter climate controls soil carbon dynamics during summer in boreal forests. *Environmental Research Letters*, 8(2), 024017. https://doi.org/10.1088/1748-9326/8/2/024017
- Hampton, S. E., Galloway, A. W., Powers, S. M., Ozersky, T., Woo, K. H., Batt, R. D., et al. (2017). Ecology under lake ice. *Ecology Letters*, 20(1), 98–111. https://doi.org/10.1111/ele.12699
- Hampton, S. E., Scheuerell, M. D., & Schindler, D. E. (2006). Coalescence in the Lake Washington story: Interaction strengths in a planktonic food web. *Limnology & Oceanography*, 51(5), 2042–2051. https://doi.org/10.4319/lo.2006.51.5.2042
- Hayden, B., Harrod, C., Thomas, S. M., Eloranta, A., Myllykangas, J.-P., Siwertsson, A., et al. (2019). From clear lakes to murky waters: Tracing the functional response of high-latitude lake communities to concurrent 'greening' and 'browning'. *Ecology Letters*, 22(5), 807–816. https:// doi.org/10.1111/ele.13238
- Hedström, P., Bystedt, D., Karlsson, J., Bokma, F., & Byström, P. (2016). Brownification increases winter mortality in fish. Oecologia, 183(2), 587–595. https://doi.org/10.1007/s00442-016-3779-y
- Helland, I. P., Finstad, A. G., Forseth, T., Hesthagen, T., & Ugedal, O. (2011). Ice-cover effects on competitive interactions between two fish species. *Journal of Animal Ecology*, 80(3), 539–547. https://doi.org/10.1111/j.1365-2656.2010.01793.x
- Ho, J. C., & Michalak, A. M. (2019). Exploring temperature and precipitation impacts on harmful algal blooms across continental U.S. lakes. Limnology & Oceanography, 65(5), 992–1009. https://doi.org/10.1002/lno.11365
- Hrycik, A., & Stockwell, J. D. (2020). Under-ice mesocosm experiments reveal the primacy of light but the importance of zooplankton in winter phytoplankton dynamics. *Limnology & Oceanography*, 66, 481–495. https://doi.org/10.1002/lno.11618
- Huisman, J., Van Oostveen, P., & Weissing, F. J. (1999). Critical depth and critical turbulence: Two different mechanisms for the development of phytoplankton blooms. *Limnology & Oceanography*, 44(7), 1781–1787. https://doi.org/10.4319/10.1999.44.7.1781
- Humphries, M. M., Studd, E. K., Menzies, A. K., & Boutin, S. (2017). To everything there is a season: Summer-to-winter food webs and the functional traits of keystone species. *Integrative and Comparative Biology*, 57(5), 961–976. https://doi.org/10.1093/icb/icx119
- Huntington, T. G., Balch, W. M., Aiken, G. R., Sheffield, J., Luo, L., Roesler, C. S., & Camill, P. (2016). Climate change and dissolved organic carbon export to the Gulf of Maine. *Journal of Geophysical Research: Biogeosciences*, 121(10), 2700–2716. https://doi.org/10.1002/2015jg003314
- Hurst, T. P. (2007). Causes and consequences of winter mortality in fishes. Journal of Fish Biology, 71(2), 315-345. https://doi.org/10.1111/j.1095-8649.2007.01596.x
- Imrit, M. A., & Sharma, S. (2021). Climate change is contributing to faster rates of lake ice loss in lakes around the Northern Hemisphere. Journal of Geophysical Research: Biogeosciences, 126. https://doi.org/10.1029/2020jg006134
- Jakkila, J., Leppäranta, M., Kawamura, T., Shirasawa, K., & Salonen, K. (2009). Radiation transfer and heat budget during the ice season in Lake Pääjärvi, Finland. Aquatic Ecology, 43(3), 681–692. https://doi.org/10.1007/s10452-009-9275-2
- Jansen, J., MacIntyre, S., Barrett, D. C., Chin, Y. P., Cortés, A., Forrest, A. L., et al. (2021). Winter limnology: How do hydrodynamics and biogeochemistry shape ecosystems under ice? *Journal of Geophysical Research: Biogeosciences*, 126(6). https://doi.org/10.1029/2020jg006237
- Jansen, J., Thornton, B. F., Jammet, M. M., Wik, M., Cortés, A., Friborg, T., et al. (2019). Climate-sensitive controls on large spring emissions of CH₄ and CO₂ from northern lakes. *Journal of Geophysical Research: Biogeosciences*, 124(7), 2379–2399. https://doi.org/10.1029/2019jg005094 Jansen, W., & Hesslein, R. H. (2004). Potential effects of climate warming on fish habitats in temperate zone lakes with special reference to Lake
- Jansen, W., & Hessieni, K. H. (2004). Forential effects of climate warming on fish naoratis in temperate zone races with spectra reference to Lake 239 of the Experimental Lakes Area (ELA), North-Western Ontario. *Environmental Biology of Fishes*, 70, 1–22. https://doi.org/10.1023/B:EB-FI.0000023035.06554.c7
- Jónasson, P. M. (1975). Population ecology and production of benthic detritivores. SIL Proceedings, 1922–2010, 19(2), 1066–1072. https://doi. org/10.1080/03680770.1974.11896157
- Jónasson, P. M. (1996). Limits for life in the lake ecosystem. SIL Proceedings, 1922–2010, 26(1), 1–33. https://doi.org/10.1080/03680770.199 5.11900691
- Joung, D., Leduc, M., Ramcharitar, B., Xu, Y., Isles, P. D., Stockwell, J. D., et al. (2017). Winter weather and lake-watershed physical configuration drive phosphorus, iron, and manganese dynamics in water and sediment of ice-covered lakes. *Limnology & Oceanography*, 62(4), 1620–1635. https://doi.org/10.1002/lno.10521
- Karpowicz, M., Zieliński, P., Grabowska, M., Ejsmont-Karabin, J., Kozłowska, J., & Feniova, I. (2020). Effect of eutrophication and humification on nutrient cycles and transfer efficiency of matter in freshwater food webs. *Hydrobiologia*, 847(11), 2521–2540. https://doi.org/10.1007/ s10750-020-04271-5
- Katz, S. L., Izmest'eva, L. R., Hampton, S. E., Ozersky, T., Shchapov, K., Moore, M. V., et al. (2015). The "Melosira years" of Lake Baikal: Winter environmental conditions at ice onset predict under-ice algal blooms in spring. *Limnology & Oceanography*, 60, 1950–1964. https:// doi.org/10.1002/lno.10143
- Kelley, D. E. (1997). Convection in ice-covered lakes: Effects on algal suspension. Journal of Plankton Research, 19(12), 1859–1880. https:// doi.org/10.1093/plankt/19.12.1859
- Kelly, P. T., Solomon, C. T., Zwart, J. A., & Jones, S. E. (2018). A framework for understanding variation in pelagic gross primary production of lake ecosystems. *Ecosystems*, 21, 1364–1376. https://doi.org/10.1007/s10021-018-0226-4
- Kerfoot, W. C., Yousef, F., Green, S. A., Budd, J. W., Schwab, D. J., & Vanderploeg, H. A. (2010). Approaching storm: Disappearing winter bloom in Lake Michigan. Journal of Great Lakes Research, 36, 30–41. https://doi.org/10.1016/j.jglr.2010.04.010
- Kienel, U., Kirillin, G., Brademann, B., Plessen, B., Lampe, R., & Brauer, A. (2017). Effects of spring warming and mixing duration on diatom deposition in deep Tiefer See, NE Germany. *Journal of Paleolimnology*, 57(1), 37–49. https://doi.org/10.1007/s10933-016-9925-z
- Kirillin, G., Leppäranta, M., Terzhevik, A., Granin, N., Bernhardt, J., Engelhardt, C., et al. (2012). Physics of seasonally ice-covered lakes: A review. Aquatic Sciences, 74(4), 659–682. https://doi.org/10.1007/s00027-012-0279-y
- Knoll, L. B., Sharma, S., Denfeld, B. A., Flaim, G., Hori, Y., Magnuson, J. J., et al. (2019). Consequences of lake and river ice loss on cultural ecosystem services. *Limnology and Oceanography Letters*, 4(5), 119–131. https://doi.org/10.1002/lol2.10116



- Lake, P. S. (2000). Disturbance, patchiness, and diversity in streams. Journal of the North American Benthological Society, 19(4), 573–592. https://doi.org/10.2307/1468118
- Lau, M. P., Valerio, G., Pilotti, M., & Hupfer, M. (2020). Intermittent meromixis controls the trophic state of warming deep lakes. Scientific Reports, 10(1). https://doi.org/10.1038/s41598-020-69721-5

Leech, D. M., Pollard, A. I., Labou, S. G., & Hampton, S. E. (2018). Fewer blue lakes and more murky lakes across the continental U.S.: Implications for planktonic food webs. *Limnology & Oceanography*, 63(6), 2661–2680. https://doi.org/10.1002/lno.10967

Leppäranta, M. (2015). Freezing of lakes and the evolution of their ice cover. Springer-Verlag Berlin An. https://doi.org/10.1007/978-3-642-29081-7 Lewis, W. M. (1983). A revised classification of lakes based on mixing. Canadian Journal of Fisheries and Aquatic Sciences, 40(10), 1779–1787. https://doi.org/10.1139/f83-207

- Likens, G. E., & Ragotzkie, R. A. (1965). Vertical water motions in a small ice-covered lake. Journal of Geophysical Research, 70(10), 2333–2344. https://doi.org/10.1029/jz070i010p02333
- Litchman, E. (1998). Population and community responses of phytoplankton to fluctuating light. *Oecologia*, 117(1-2), 247-257. https://doi.org/10.1007/s004420050655
- Litchman, E., & Klausmeier, C. A. (2008). Trait-based community ecology of phytoplankton. Annual Review of Ecology, Evolution, and Systematics, 39(1), 615–639. https://doi.org/10.1146/annurev.ecolsys.39.110707.173549
- Loeks-Johnson, B. M., & Cotner, J. B. (2020). Upper Midwest lakes are supersaturated with N₂. Proceedings of the National Academy of Sciences, 117(29), 17063–17067. https://doi.org/10.1073/pnas.1921689117
- Lomas, M. W., & Glibert, P. M. (1999). Interactions between NH₄ and NO₃ uptake and assimilation: Comparison of diatoms and dinoflagellates at several growth temperatures. *Marine Biology*, 133, 541–551. https://doi.org/10.1007/s002270050494
- MacIntyre, S., Crowe, A. T., Cortés, A., & Arneborg, L. (2018). Turbulence in a small arctic pond. *Limnology & Oceanography*, 63(6), 2337–2358. https://doi.org/10.1002/lno.10941
- Magnuson, J. J., Beckel, A. L., Mills, K., & Brandt, S. B. (1985). Surviving winter hypoxia: Behavioral adaptations of fishes in a northern Wisconsin winterkill lake. *Environmental Biology of Fishes*, 14(4), 241–250. https://doi.org/10.1007/bf00002627
- Magnuson, J. J., Robertson, D. M., Benson, B. J., Wynne, R. H., Livingstone, D. M., Arai, T., et al. (2000). Historical trends in lake and river ice cover in the Northern Hemisphere. Science, 289(5485), 1743–1746. https://doi.org/10.1126/science.289.5485.1743

Malm, J. (1998). Bottom buoyancy layer in an ice-covered lake. *Water Resources Research*, 34(11), 2981–2993. https://doi.org/10.1029/98wr01904 Marchand, P. J. (1987). *Life in the cold: An introduction to winter ecology*. University Press of New England. https://doi.org/10.2307/1310957

- Markelov, I., Couture, R. M., Fischer, R., Haande, S., & Van Cappellen, P. (2019). Coupling water column and sediment biogeochemical dynamics: Modeling internal phosphorus loading, climate change responses, and mitigation measures in Lake Vansjø, Norway. Journal of Geophysical Research: Biogeosciences, 124(12), 3847–3866. https://doi.org/10.1029/2019JG005254
- Massé, S., Botrel, M., Walsh, D. A., & Maranger, R. (2019). Annual nitrification dynamics in a seasonally ice-covered lake. PLoS One, 14(3), 1–21. https://doi.org/10.1371/journal.pone.0213748
- McKnight, D. M., Aiken, G. R., Andrews, E. D., Bowles, E. C., & Harnish, R. A. (1993). Dissolved organic material in dry valley lakes: A comparison of Lake Fryxell, Lake Hoare and Lake Vanda. *Physical and Biogeochemical Processes in Antarctic Lakes Antarctic Research Series*, 59, 119–133. https://doi.org/10.1029/ar059p0119
- McKnight, D. M., Howes, B. L., Taylor, C. D., & Goehringer, D. D. (2000). Phytoplankton dynamics in a stable stratified Antarctic lake during winter darkness. *Journal of Phycology*, 36(5), 852–861. https://doi.org/10.1046/j.1529-8817.2000.00031.x
- McMeans, B. C., Mccann, K. S., Guzzo, M. M., Bartley, T. J., Bieg, C., Blanchfield, P. J., et al. (2020). Winter in water: Differential responses and the maintenance of biodiversity. *Ecology Letters*, 23(6), 922–938. https://doi.org/10.1111/ele.13504
- McMeans, B. C., McCann, K. S., Humphries, M., Rooney, N., & Fisk, A. T. (2015). Food deb structure in temporally-forced ecosystems. Trends in Ecology & Evolution, 30(11), 662–672. https://doi.org/10.1016/j.tree.2015.09.001
- Melack, J. M., Sadro, S., Sickman, J. O., & Dozier, J. (2020). Lakes and watersheds in the Sierra Nevada of California responses to environmental change. University of California Press. https://doi.org/10.2307/j.ctv17hm9sr
- Missaghi, S., Hondzo, M., & Herb, W. (2017). Prediction of lake water temperature, dissolved oxygen, and fish habitat under changing climate. *Climatic Change*, 141(4), 747–757. https://doi.org/10.1007/s10584-017-1916-1
- Mortimer, C. H., & Mackereth, F. J. (1958). Convection and its consequences in ice-covered lakes. SIL Proceedings, 1922–2010, 13(2), 923–932. https://doi.org/10.1080/03680770.1956.11895490
- North, R. P., North, R. L., Livingstone, D. M., Köster, O., & Kipfer, R. (2014). Long-term changes in hypoxia and soluble reactive phosphorus in the hypolimnion of a large temperate lake: Consequences of a climate regime shift. *Global Change Biology*, 20(3), 811–823. https://doi. org/10.1111/gcb.12371
- Nürnberg, G. K., Molot, L. A., O'Connor, E., Jarjanazi, H., Winter, J., & Young, J. (2013). Evidence for internal phosphorus loading, hypoxia and effects on phytoplankton in partially polymictic Lake Simcoe, Ontario. *Journal of the Great Lake Research*, 39, 259–270. https://doi. org/10.1016/j.jglr.2013.03.016
- Obrador, B., Staehr, P. A., & Christensen, J. P. C. (2014). Vertical patterns of metabolism in three contrasting stratified lakes. *Limnology & Oceanography*, 59(4), 1228–1240. https://doi.org/10.4319/lo.2014.59.4.1228
- Olson, M. H., Fischer, J. M., Hayashi, M., & Williamson, C. E. (2020). Meteorological drivers of interannual variation in transparency of mountain lakes. Arctic, Antarctic, and Alpine Research, 52(1), 424–434. https://doi.org/10.1080/15230430.2020.1800972
- Orihel, D. M., Baulch, H. M., Casson, N. J., North, R. L., Parsons, C. T., Seckar, D. C. M., & Venkiteswaran, J. J. (2017). Internal phosphorus loading in Canadian fresh waters: A critical review and data analysis. *Canadian Journal of Fisheries and Aquatic Sciences*, 74(12), 2005–2029. https://doi.org/10.1139/cjfas-2016-0500
- Özkundakci, D., Gsell, A. S., Hintze, T., Täuscher, H., & Adrian, R. (2016). Winter severity determines functional trait composition of phytoplankton in seasonally ice-covered lakes. *Global Change Biology*, 22(1), 284–298. https://doi.org/10.1111/gcb.13085
- Palacin-Lizarbe, C., Camarero, L., & Catalan, J. (2018). Denitrification temperature dependence in remote, cold and N-poor lake sediments. Water Resources Research, 54(2), 1161–1173. https://doi.org/10.1002/2017WR021680
- Pasztaleniec, A., & Lenard, T. (2008). Winter phytoplankton communities in different depths of three mesotrophic lakes (Łęczna-Włodawa Lakeland, Eastern Poland). *Biologia*, 63(3), 294–301. https://doi.org/10.2478/s11756-008-0062-7
- Pennak, R. W. (1968). Field and experimental winter limnology of three Colorado mountain lakes. *Ecology*, 49, 505–520. https://doi.org/10.2307/1934117
- Perga, M., Syarki, M., Spangenberg, J. E., Frossard, V., Lyautey, E., Kalinkina, N., & Bouffard, D. (2020). Fasting or feeding: A planktonic food web under lake ice. *Freshwater Biology*, 66, 570–581. https://doi.org/10.1111/fwb.13661
- Perovich, D. K. (2007). Light reflection and transmission by a temperate snow cover. Journal of Glaciology, 53(181), 201–210. https://doi. org/10.3189/172756507782202919



- Powers, S. M., Baulch, H. M., Hampton, S. E., Labou, S. G., Lottig, N. R., & Stanley, E. H. (2017). Nitrification contributes to winter oxygen depletion in seasonally frozen forested lakes. *Biogeochemistry*, 136(2), 119–129. https://doi.org/10.1007/s10533-017-0382-1
- Powers, S. M., Labou, S. G., Baulch, H. M., Hunt, R. J., Lottig, N. R., Hampton, S. E., & Stanley, E. H. (2017). Ice duration drives winter nitrate accumulation in north temperate lakes. *Limnology & Oceanography Letters*, 2(5), 177–186. https://doi.org/10.1002/lol2.10048
- Preston, D. L., Caine, N., Mcknight, D. M., Williams, M. W., Hell, K., Miller, M. P., et al. (2016). Climate regulates alpine lake ice cover phenology and aquatic ecosystem structure. *Geophysical Research Letters*, 43(10), 5353–5360. https://doi.org/10.1002/2016GL069036
- Prowse, T. D., & Stephenson, R. L. (1986). The relationship between winter lake cover, radiation receipts and the oxygen deficit in temperate lakes. Atmosphere - Ocean, 24(4), 386–403. https://doi.org/10.1080/07055900.1986.9649259
- Rabaey, J. S., Domine, L. M., Zimmer, K. D., & Cotner, J. B. (2021). Winter oxygen regimes in clear and turbid shallow lakes. *Journal of Geophysical Research: Biogeosciences*, 126(3). https://doi.org/10.1029/2020jg006065
- Rattan, K., Corriveau, J., Brua, R., Culp, J., Yates, A., & Chambers, P. (2017). Quantifying seasonal variation in total phosphorus and nitrogen from prairie streams in the Red River Basin, Manitoba Canada. Science of the Total Environment, 575, 649–659. https://doi.org/10.1016/j.scitotenv.2016.09.073
- Rautio, M., Imbeau, E., & Vincent, W. F. (2021). Lake Simoncouch: Particulate matter and dissolved organic carbon concentrations in northern lake ice with associated limnological observations (The Lake Ice Continuum Concept), v. 1.0 (2017–2018). Nordicana D90. https://doi. org/10.5885/45717CE-5A9BFDF1C9064936
- Rautio, M., Mariash, H., & Forsström, L. (2011). Seasonal shifts between autochthonous and allochthonous carbon contributions to zooplankton diets in a subarctic lake. *Limnology & Oceanography*, 56(4), 1513–1524. https://doi.org/10.4319/lo.2011.56.4.1513
- Raymond, P. A., Saiers, J. E., & Sobczak, W. V. (2016). Hydrological and biogeochemical controls on watershed dissolved organic matter transport: Pulse-shunt concept. *Ecology*, 97(1), 5–16. https://doi.org/10.1890/14-1684.1
- Roiha, T., Laurion, I., & Rautio, M. (2015). Carbon dynamics in highly heterotrophic subarctic thaw ponds. *Biogeosciences Discussions*, 12(14), 11707–11749. https://doi.org/10.5194/bgd-12-11707-2015
- Rowe, M. D., Anderson, E. J., Vanderploeg, H. A., Pothoven, S. A., Elgin, A. K., Wang, J., & Yousef, F. (2017). Influence of invasive quagga mussels, phosphorus loads, and climate on spatial and temporal patterns of productivity in Lake Michigan: A biophysical modeling study. *Limnology & Oceanography*, 62(6), 2629–2649. https://doi.org/10.1002/lno.10595
- Rue, G. P., Darling, J. P., Graham, E., Tfaily, M. M., & Mcknight, D. M. (2019). Dynamic changes in dissolved organic matter composition in a Mountain Lake under ice cover and relationships to changes in nutrient cycling and phytoplankton community composition. Aquatic Sciences, 82(1). https://doi.org/10.1007/s00027-019-0687-3
- Sadro, S., Melack, J. M., Sickman, J. O., & Skeen, K. (2018). Climate warming response of mountain lakes affected by variations in snow. *Limnology & Oceanography Letters*, 4(1), 9–17. https://doi.org/10.1002/lo12.10099
- Salonen, K., Leppäranta, M., Viljanen, M., & Gulati, R. D. (2009). Perspectives in winter limnology: Closing the annual cycle of freezing lakes. Aquatic Ecology, 43(3), 609–616. https://doi.org/10.1007/s10452-009-9278-z
- Santibáñez, P. A., Michaud, A. B., Vick-Majors, T. J., D'Andrilli, J., Chiuchiolo, A., Hand, K. P., & Priscu, J. C. (2019). Differential incorporation of bacteria, organic matter, and inorganic ions into lake ice during ice formation. *Journal of Geophysical Research: Biogeosciences*, 124(3), 585–600. https://doi.org/10.1029/2018jg004825
- Schindler, D. W., & Smol, J. P. (2006). Cumulative effects of climate warming and other human activities on freshwaters of Arctic and Subarctic North America. AMBIO: A Journal of the Human Environment, 35(4), 160–168. https://doi.org/10.1579/0044-7447(2006)35[160:ceocwa]2.0.co;2
- Schneider, T., Grosbois, G., Vincent, W. F., & Rautio, M. (2017). Saving for the future: Pre-winter uptake of algal lipids supports copepod egg production in spring. *Freshwater Biology*, 62(6), 1063–1072. https://doi.org/10.1111/fwb.12925
- Schütte, U. M., Cadieux, S. B., Hemmerich, C., Pratt, L. M., & White, J. R. (2016). Unanticipated geochemical and microbial community structure under seasonal ice cover in a dilute, dimictic Arctic lake. Frontiers in Microbiology, 7, 1035. https://doi.org/10.3389/fmicb.2016.01035
- Seastedt, T. R., Bowman, W. D., Caine, T. N., Mcknight, D., Townsend, A., & Williams, M. W. (2004). The Landscape Continuum: A model for high-elevation ecosystems. *BioScience*, 54(2), 111. https://doi.org/10.1641/0006-3568(2004)054[0111:tlcamf]2.0.co;2
- Sharma, S., Blagrave, K., Magnuson, J. J., O'Reilly, C. M., Oliver, S., Batt, R. D., et al. (2019). Widespread loss of lake ice around the Northern Hemisphere in a warming world. *Nature Climate Change*, 9(3), 227–231. https://doi.org/10.1038/s41558-018-0393-5
- Shatwell, T., Köhler, J., & Nicklisch, A. (2008). Warming promotes cold-adapted phytoplankton in temperate lakes and opens a loophole for Oscillatoriales in spring. *Global Change Biology*, 14(9), 2194–2200. https://doi.org/10.1111/j.1365-2486.2008.01630.x
- Shatwell, T., Thiery, W., & Kirillin, G. (2019). Future projections of temperature and mixing regime of European temperate lakes. *Hydrology and Earth System Science*, 23(3), 1533–1551. https://doi.org/10.5194/hess-23-1533-2019
- Shuter, B. J., Finstad, A. G., Helland, I. P., Zweimüller, I., & Hölker, F. (2012). The role of winter phenology in shaping the ecology of freshwater fish and their sensitivities to climate change. *Aquatic Sciences*, 74(4), 637–657. https://doi.org/10.1007/s00027-012-0274-3
- Shuter, B. J., Giacomini, H. C., De Kerckhove, D., & Vascotto, K. (2016). Fish life history dynamics: Shifts in prey size structure evoke shifts in predator maturation traits. *Canadian Journal of Fisheries and Aquatic Sciences*, 73(4), 693–708. https://doi.org/10.1139/cjfas-2015-0190
- Shuter, B. J., & Ing, K. K. (1997). Factors affecting the production of zooplankton in lakes. *Canadian Journal of Fisheries and Aquatic Sciences*, 54(2), 359–377. https://doi.org/10.1139/f96-270
- Sickman, J. O., Melack, J. M., & Clow, D. W. (2003). Evidence for nutrient enrichment of high-elevation lakes in the Sierra Nevada, California. Limnology & Oceanography, 48(5), 1885–1892. https://doi.org/10.4319/lo.2003.48.5.1885
- Smits, A. P., Macintyre, S., & Sadro, S. (2020). Snowpack determines the relative importance of climate factors driving summer lake warming. Limnology & Oceanography Letters, 5(3), 271–279. https://doi.org/10.1002/lol2.10147
- Sobek, S., Durisch-Kaiser, E., Zurbrügg, R., Wongfun, N., Wessels, M., Pasche, N., & Wehrli, B. (2009). Organic carbon burial efficiency in lake sediments controlled by oxygen exposure time and sediment source. *Limnology & Oceanography*, 54(6), 2243–2254. https://doi.org/10.4319/ lo.2009.54.6.2243
- Solomon, C. T., Jones, S. E., Weidel, B. C., Buffam, I., Fork, M. L., Karlsson, J., et al. (2015). Ecosystem consequences of changing inputs of terrestrial dissolved organic matter to lakes: Current knowledge and future challenges. *Ecosystems*, 18(3), 376–389. https://doi.org/10.1007/ s10021-015-9848-y
- Sommer, U., Adrian, R., De Senerpont Domis, L., Elser, J. J., Gaedke, U., Ibelings, B., et al. (2012). Beyond the Plankton Ecology Group (PEG) model: Mechanisms driving plankton succession. *Annual Review of Ecology, Evolution, and Systematics*, 43(1), 429–448. https://doi. org/10.1146/annurev-ecolsys-110411-160251
- Speers-Roesch, B., Norin, T., & Driedzic, W. R. (2018). The benefit of being still: Energy savings during winter dormancy in fish come from inactivity and the cold, not from metabolic rate depression. *Proceedings of the Royal Society B: Biological Sciences*, 285(1886), 20181593. https://doi.org/10.1098/rspb.2018.1593



- Staehr, P. A., Sand-Jensen, K., Raun, A. L., Nilsson, B., & Kidmose, J. (2010). Drivers of metabolism and net heterotrophy in contrasting lakes. Limnology & Oceanography, 55(2), 817–830. https://doi.org/10.4319/lo.2010.55.2.0817
- Stockner, J. G., & Porter, K. G. (1988). Microbial food webs in freshwater planktonic ecosystems. In S. R. Carpenter (Ed.), Complex interactions in lake communities (pp. 69–83). Springer-Verlag New York Inc. https://doi.org/10.1007/978-1-4612-3838-6_5
- Stoecker, D. K. (1998). Conceptual models of mixotrophy in planktonic protists and some ecological and evolutionary implications. *European Journal of Protistology*, 34(3), 281–290. https://doi.org/10.1016/s0932-4739(98)80055-2
- Strayer, D. L. (2009). Twenty years of zebra mussels: Lessons from the mollusk that made headlines. *Frontiers in Ecology and the Environment*, 7(3), 135–141. https://doi.org/10.1890/080020
- Sutcliffe, D. W., Carrick, T. R., & Willoughby, L. G. (1981). Effects of diet, body size, age and temperature on growth rates in the amphipod Gammarus pulex. Freshwater Biology, 11(2), 183–214. https://doi.org/10.1111/j.1365-2427.1981.tb01252.x
- Thackeray, S. J., Sparks, T. H., Frederiksen, M., Burthe, S., Bacon, S. J., Bell, J. R., et al. (2010). Trophic level asynchrony in rates of phenological change for marine, freshwater and terrestrial environments. *Global Change Biology*, 16(12), 3304–3313. https://doi.org/10.1111/j.1365-2486.2010.02165.x
- Thompson, S., & Cotner, J. (2020). P-limitation drives changes in DOM production by aquatic bacteria. Aquatic Microbial Ecology, 85, 35–46. https://doi.org/10.3354/ame01940
- Thompson, S. K., & Cotner, J. B. (2018). Bioavailability of dissolved organic phosphorus in temperate lakes. Frontiers in Environmental Science, 6. https://doi.org/10.3389/fenvs.2018.00062
- Timoshkin, O. A. (2001). Lake Baikal: Faunal diversity, problem of its immiscibility and origin, ecology, and "Exotic" communities. In Annotirovannyi spisok fauny ozera Baikal i ego vodosbornogo basseina. Spravochniki i opredeliteli po faune i flore ozera Baikal. T. 1. Ozero Baikal (Annotated faunal list of Lake Baikal and its drainage basin. Manuals and guidebooks on the fauna and flora of Lake Baikal. Vol. 1. Lake Baikal). Nauka.
- Tõnno, I., Ott, K., & Nõges, T. (2005). Nitrogen dynamics in the steeply stratified, temperate Lake Verevi, Estonia. Hydrobiologia, 547(1), 63–71. https://doi.org/10.1007/s10750-005-4145-y
- Tran, P., Ramachandran, A., Khawasik, O., Beisner, B. E., Rautio, M., Huot, Y., & Walsh, D. A. (2018). Microbial life under ice: Metagenome diversity and in-situ activity of Verrucomicrobia in seasonally ice-covered Lakes. *Environmental Microbiology*, 20(7), 2568–2584. https://doi. org/10.1111/1462-2920.14283
- Twiss, M. R., McKay, R. M. L., Bourbonniere, R. A., Bullerjahn, G. S., Carrick, H. J., Smith, R. E. H., et al. (2012). Diatoms abound in ice-covered Lake Erie: An investigation of offshore winter limnology in Lake Erie over the period 2007 to 2010. Journal of Great Lakes Research, 38(1), 18–30. https://doi.org/10.1016/j.jglr.2011.12.008
- Urban, N. R., Brezonik, P. L., Baker, L. A., & Sherman, L. A. (1994). Sulfate reduction and diffusion in sediments of Little Rock Lake, Wisconsin. Limnology & Oceanography, 39(4), 797–815. https://doi.org/10.4319/10.1994.39.4.0797
- Vad, C. F., Schneider, C., Lukić, D., Horváth, Z., Kainz, M. J., Stibor, H., & Ptacnik, R. (2020). Grazing resistance and poor food quality of a widespread mixotroph impair zooplankton secondary production. *Oecologia*, 193(2), 489–502. https://doi.org/10.1007/s00442-020-04677-x
- Vanderploeg, H. A., Liebig, J. R., Nalepa, T. F., Fahnenstiel, G. L., & Pothoven, S. A. (2010). Dreissena and the disappearance of the spring phytoplankton bloom in Lake Michigan. Journal of Great Lakes Research, 36, 50–59. https://doi.org/10.1016/j.jglr.2010.04.005
- Vannote, R. L., Minshall, G. W., Cummins, K. W., Sedell, J. R., & Cushing, C. E. (1980). The river continuum concept. Canadian Journal of Fisheries and Aquatic Sciences, 37(1), 130–137. https://doi.org/10.1139/f80-017
- Varpe, Ø., Daase, M., & Kristiansen, T. (2015). A fish-eye view on the new Arctic lightscape. ICES Journal of Marine Science, 72(9), 2532–2538. https://doi.org/10.1093/icesjms/fsv129
- Verpoorter, C., Kutser, T., Seekell, D. A., & Tranvik, L. J. (2014). A global inventory of lakes based on high-resolution satellite imagery. Geophysical Research Letters, 41(18), 6396–6402. https://doi.org/10.1002/2014gl060641
- Vigneron, A., Lovejoy, C., Cruaud, P., Kalenitchenko, D., Culley, A., & Vincent, W. F. (2019). Contrasting winter versus summer microbial communities and metabolic functions in a permafrost thaw lake. *Frontiers in Microbiology*, 10. https://doi.org/10.3389/fmicb.2019.01656
- Vincent, W. F. (1983). Phytoplankton production and winter mixing: Contrasting effects in two oligotrophic lakes. Journal of Ecology, 71(1), 1. https://doi.org/10.2307/2259960
- Vörös, L., Mózes, A., & Somogyi, B. (2009). A five-year study of autotrophic winter picoplankton in Lake Balaton, Hungary. Aquatic Ecology, 43(3), 727–734. https://doi.org/10.1007/s10452-009-9272-5
- Ward, J. V., & Stanford, J. A. (1982). The serial discontinuity concept of lotic ecosystems. In T. D. Fontaine, & S. M. Bartell (Eds.), Dynamics of lotic ecosystems (Vol. 17, pp. 29–310). Science. https://doi.org/10.1002/rrr.659
- Warren, S. G. (2019). Optical properties of ice and snow. Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences, 377(2146), 20180161. https://doi.org/10.1098/rsta.2018.0161
- Watson, B. M., Biagi, C. A., Northrup, S. L., Ohata, M. L., Charles, C., Blanchfield, P. J., et al. (2019). Distinct diel and seasonal behaviours in rainbow trout detected by fine-scale acoustic telemetry in a lake environment. *Canadian Journal of Fisheries and Aquatic Sciences*, 76(8), 1432–1445. https://doi.org/10.1139/cjfas-2018-0293
- Weyhenmeyer, G. A., Westöö, A., & Willén, E. (2008). Increasingly ice-free winters and their effects on water quality in Sweden's largeest lakes. *European Large Lakes Ecosystem Changes and Their Ecological and Socioeconomic Impacts*, 111–118. https://doi. org/10.1007/978-1-4020-8379-2_13
- Woolway, R. I., Kraemer, B. M., Lenters, J. D., Merchant, C. J., O'Reilly, C. M., & Sharma, S. (2020). Global lake responses to climate change. *Nature Reviews Earth & Environment*, 1(8), 388–403. https://doi.org/10.1038/s43017-020-0067-5
- Woolway, R. I., & Merchant, C. J. (2019). Worldwide alteration of lake mixing regimes in response to climate change. *Nature Geoscience*, *12*(4), 271–276. https://doi.org/10.1038/s41561-019-0322-x
- Xiao, K., Griffis, T. J., Baker, J. M., Bolstad, P. V., Erickson, M. D., Lee, X., et al. (2018). Evaporation from a temperate closed-basin lake and its impact on present, past, and future water level. *Journal of Hydrology*, 561, 59–75. https://doi.org/10.1016/j.jhydrol.2018.03.059
- Yang, B., Wells, M. G., Li, J., & Young, J. (2020). Mixing, stratification, and plankton under lake-ice during winter in a large lake: Implications for spring dissolved oxygen levels. *Limnology & Oceanography*, 65, 2713–2729. https://doi.org/10.1002/lno.11543
- Yang, B., Wells, M. G., McMeans, B. C., Dugan, H. A., Rusak, J. A., Weyhenmeyer, G. A., et al. (2021). A new thermal categorization of ice-covered lakes. *Geophysical Research Letters*, 48, e2020GL091374. https://doi.org/10.1029/2020GL091374
- Yang, B., Young, J., Brown, L., & Wells, M. (2017). High-frequency observations of temperature and dissolved oxygen reveal under-ice convection in a large lake. *Geophysical Research Letters*, 44(24), 12218–12226. https://doi.org/10.1002/2017GL075373
- Yankova, Y., Neuenschwander, S., Köster, O., & Posch, T. (2017). Abrupt stop of deep water turnover with lake warming: Drastic consequences for algal primary producers. *Scientific Reports*, 7(1). https://doi.org/10.1038/s41598-017-13159-9